Contents lists available at ScienceDirect





journal homepage: www.elsevier.com/locate/pacfin

China's financial network with international spillovers: A first look

Jian Yang^{a,*}, Ziliang Yu^b, Jun Ma^c

^a Center for China Financial Research, Business School, University of Colorado Denver, United States of America

^b School of Finance, Nankai University, China

^c Center for Finance and Development, Tsinghua University, China

ARTICLE INFO

Keywords: China's financial system Shadow banking Systemic risk Macroprudential regulation Financial network

JEL classification: G01 G15

ABSTRACT

Using a modified spillover index approach from the perspective of financial shocks transmission, this study is the first to explore China's financial institution (FI) network after the global financial crisis, allowing for interactions with the financial sectors of four major global economies. We document that: (1) although banks still dominate China's financial sector, nonbank FIs also bear considerable influence; (2) the market-oriented large commercial banks generally play a more pronounced role than the four state-owned megabanks in transmitting financial shocks; (3) China's financial sector exerts noticeable influence on the global financial sector, particularly that of Japan; and (4) monetary policy measures dominate in determining the overall influence from other FIs to a particular FI while firm-specific factors dominate in determining the influence of a particular FI on other FIs. These findings have important policy implications.

1. Introduction

The 2015–2016 Chinese stock market turbulence triggered global fears over the possibility of another global crisis.¹ The worldwide anxiety again underscored the conventional wisdom prevailing since the 2008 global financial crisis: understanding the mechanism of financial shock transmission among financial institutions (FIs) is crucial to prevent the occurrence and propagation of financial crises, establish efficient regulation and supervision, and promote appropriate asset pricing and risk management (Acharya et al., 2012, 2017; Acemoglu et al., 2012, 2015; Adrian and Brunnermeier, 2016). Unfortunately, despite basic facts underscoring the global importance of China's financial system—China has the second-largest stock market in the world and the four biggest state-owned Chinese banks (i.e., the Big Four—the Industrial and Commercial Bank of China, the Bank of China, the China Construction Bank, and the Agricultural Bank of China) are among the top ten FIs in the world—there is no comprehensive study on the financial system. This study attempts to fill this gap.

Based on a modified financial network analysis (Diebold and Yilmaz, 2014; Yang and Zhou, 2013), we investigate the network structure and potential determinants of financial shock transmission among China's FIs since the onset of the 2008 global financial crisis, while controlling for the interactions between China and the four countries with the largest global financial services sectors (i.e., the United States [US], the United Kingdom [UK], Germany, and Japan). Similar to Diebold and Yilmaz, 2014 and Yang and

E-mail addresses: jian.yang@ucdenver.edu (J. Yang), yuziliang@nankai.edu.cn (Z. Yu), maj@pbcsf.tsinghua.edu.cn (J. Ma).

https://doi.org/10.1016/j.pacfin.2019.101222

Received 26 February 2019; Received in revised form 19 September 2019; Accepted 14 October 2019 Available online 31 October 2019 0927-538X/ © 2019 Elsevier B.V. All rights reserved.





^{*} Corresponding author.

¹ During the turbulence, the Shanghai stock market had fallen 30% within a month (by July 9, 2015), and eventually lost 50% until the market became tranquil in February 2016. China's stock market slump in 2015 dominated discussions at the October 2015 International Monetary Fund (IMF) annual meeting of global finance ministers and central bankers held in Peru, with participants asking whether "China's economic downturn [would] trigger a new financial crisis." Interestingly, Allen et al. (2012) also suggested that China should be vigilant against a "twin crisis" consisting of simultaneous foreign exchange and banking/stock market crises, which would impair sustainable economic growth in China.

J. Yang, et al.

Zhou (2013), we define the systemically important financial institutions (SIFIs) as those having relatively more influence, and thus, positive net influence (influence on others minus influence from others) on other institutions in the financial network. The influence of an FI on other FIs and its net influence in the financial shock transmission network arguably reflect the comparative importance of an FI within the network.

As reviewed below, our study generally falls into the large emerging body of literature on identifying SIFIs using public market data. In this study, we use the stock returns,² based on a modified approach to the recently developed financial network analysis, to investigate China's financial shocks transmission network and the SIFIs in China, rather than other popular systemic risk measures used in previous studies.³ This is so because all the current major measures on systemic risk mirror ranking of firms based on market risk or liabilities, which are (largely) reflected in the stock prices (Benoit et al., 2013, 2017).

Our study is particularly similar to Yang and Zhou (2013) and Diebold and Yilmaz, 2014. Yang and Zhou (2013) use credit default swap data to identify the structure of credit risk network across the major US and EU FIs. Diebold and Yilmaz, 2014 use stock returns to investigate the network connectedness among major US FIs based on the proposed network analysis derived from the vector autogresression (VAR) forecast error variance decomposition. Our study can be regarded as using an analytical approach combining the approaches used in these two studies, with special focus on Chinese listed FIs. First, we use the stock returns to investigate the financial transmission network among Chinese listed FIs, based on the modified Diebold and Yilmaz, 2014 network analysis. Following the two-step analytical approach proposed in Yang and Zhou (2013), we then further investigate the relevant determinants of such a network.

The study contributes in two ways. First, we contribute to a further understanding of China's financial system. Currently, the literature argues that China's financial system is still dominated by banks, especially by the Big Four (Allen et al., 2005; Berger et al., 2009; Ayyagari et al., 2010; Allen et al., 2012). As against this, we document a new finding that although banks still dominate China's financial system in terms of transmitting financial shocks, nonbank FIs' shocks already bear considerable influence on banks. This finding provides additional evidence on the importance of China's shadow banking problems during recent years (e.g., Allen et al., 2012; Tobin and Volz, 2018; Yang et al., 2019). Consistent with the importance of nonbank FIs and shadow banking problems, we also document the first empirical evidence that insurance companies in China largely resemble commercial banks on the basis of their stock market performance during the sample period.

We also present new evidence that after the global financial crisis, China's financial sector has surprisingly exerted considerable influence on the financial sectors in the four major developed countries. This influence is especially evident in the Japanese financial sector. Such a finding is intriguing, given the well-documented low correlations between Chinese and major global stock markets, especially between China and Japan (Carpenter et al., 2015; Jach, 2017), and the fact that China's financial system is "centrally controlled, bank-dominated, uniquely relationship-driven, [...], rather than based primarily on securities markets and legal contracts" (Carpenter and Whitelaw, 2017).⁴

Second, we contribute to the growing literature on systemic risk by exploring the transmission network among China's FIs while controlling for the influence from the financial sectors of major economies. Despite the fact that since 2008, four of the ten largest FIs in the world are Chinese, we are the first (to our best knowledge) to attempt a comprehensive examination of the pattern and determinants of financial shock transmission in China. Accemoglu et al. (2015, p. 564) argue that "the exact role played by the financial system's architecture in creating systemic risk remains, at best, imperfectly understood." The argument is even stronger in the case of China, given the unique features of its financial system, ⁵ as pointed out in Carpenter and Whitelaw (2017).

We document a striking new finding that the market-oriented large commercial banks often play a more pronounced role than the Big Four in the financial shock transmission network, despite the latter's predominance in China's banking system. However, the role of these FIs is not static but changes quite dramatically over time. Interestingly, the Big Four do become relatively more influential in terms of financial shock transmission, primarily during turbulent periods (the 2008 financial crisis and the 2015 Chinese stock market crash), compared to tranquil periods. Further, extending many earlier studies (e.g., Yang and Zhou, 2013; Ballester et al., 2016; Helwege and Zhang, 2015), we find that various macroeconomic factors, especially China's monetary policy measures (including the money supply, interbank lending rate, and exchange rate), dominate in determining the influence of others on a particular FI in China. Meanwhile, we also find that firm-specific factors (e.g., leverage, size, and so on) dominate in determining the influence of an FI on other entities (as well as the net influence) in the network of financial shock transmission. These findings are supportive of the argument that microprudential regulation and supervision based on the conventional firm-specific approach are particularly insufficient to ensure financial stability in emerging economies, as underscored by Hahm et al. (2012).

² As discussed in more detail later in the literature review, recent studies such as Carpenter et al. (2015) suggest good informational quality of stock prices of Chinese listed companies.

³ These are the Marginal Expected Shortfall and the Systemic Expected Shortfall of Acharya et al. (2012), the Systemic Risk Measure of Acharya et al. (2017) and Brownlees and Engle (2016), and the Delta Conditional Value-at-Risks of Adrian and Brunnermeier (2016).

⁴ The above features could imply that China's financial sector might be informationally lagging or even still largely segmented from the rest of the world. The existing literature also documents macroeconomic spillover only from the US to China (e.g., Pang and Siklos, 2016) but little from China to the US. However, as correctly pointed out by Carpenter and Whitelaw (2017), we should avoid over-applying research findings developed for the US setting to understand China's distinctive financial system.

⁵ Take Chinese stock markets as an example. Such special features include 1-day minimum holding period, a 10% daily price move limit, short-sale restriction, trading suspension, IPO suspension, direct government intervention, and special treatment status for distressed stocks, as well as non-tradable shares, market segmentation, and limited institutional participation (Carpenter and Whitelaw, 2017). All these unique features point to the real possibility that the findings on the US and other developed countries might or might not apply in China's setting.

The rest of this study is organized as follows. Section 2 reviews the literature. Section 3 describes the data. Section 4 discusses the empirical methodology. Section 5 presents the empirical results. Section 6 further explores various determinants. Section 7 concludes the study.

2. Literature review

This section provides a brief literature review to shed some light on why FIs in China may play an important role in the transmission of financial shocks.

According to the classification of Benoit et al. (2013, 2017), there is a large emerging body of literature on measuring systemic risk and identifying SIFIs using public market data, although there is an alternative approach to identify SIFIs by relying on information on positions and risk exposures. The high-frequency public market data, such as stock returns, option prices, or credit default swap spreads should reflect all information about publicly traded firms, including publicly traded FIs. Thus, using public market data should be an efficient approach to investigate the up-to-date risk transmission network as well as identify SIFIs (Huang et al., 2009; Benoit et al., 2013, 2017; Yang and Zhou, 2013; Diebold and Yilmaz, 2014).

In this regard, one might be concerned about the informational quality of stock prices on the Chinese stock market due to its unique features, although it has consistently ranked as the second-largest stock market since 2014. Hence, it is important to note that although Chinese stock prices are more volatile, Carpenter et al. (2015), among others, recently found that "since the reforms of the last decade, China's stock market has become as informative about future corporate profits as the US. Moreover, though it is a segmented market, Chinese investors price risk and other stock characteristics remarkably like investors in other large economies." Furthermore, these listed FIs are generally among the largest and the most actively traded on the Chinese stock market, further strengthening the evidence on informational quality.

From another perspective, there are also various empirical approaches to measure systemic risk that carry direct implications for risk transmission. These approaches include financial index methods (e.g., IMF, BIS, and FSB, 2009; Allahrakha et al., 2015; Glasserman and Loudis, 2015), structural methods based on asset-liability and interbank market data (e.g., Mistrulli, 2011), and the reduced-form approach based on financial market data (e.g., Adrian and Brunnermeier, 2016; Acharya et al., 2012, 2017). The empirical approach adopted in this study is a reduced-form approach similar to Diebold and Yilmaz, 2014 and Yang and Zhou (2013), which can better model the interconnectedness of FIs or risk transmission beyond the tangible business connections. While not without its own limitations, such a capacity to comprehensively capture systemic risk should be valuable, because systemic risk does come from various sources beyond tangible business connections (Benoit et al., 2013, 2017).

On the theoretical dimensions, there may be various considerations or models that can motivate systemic risk and their transmission, where we use financial shocks more or less as a proxy for systemic risk. Allen et al. (2009) point out that there are at least three types of systemic risk that have direct implications for risk transmission among various FIs. Specifically, the first is a common asset shock (e.g., a fall in real estate or stock market prices), while the second may be a contagion where the failure of one FI leads to the failure of another due to investor panics or other psychological factors. The third common type of systemic risk is the failure of one FI that coincides with the failure of many others due to highly correlated portfolios among individual FIs. While Benoit et al. (2017) also discussed largely similar channels of systemic risk transmissions among FIs (e.g., systemic risk-taking through business operations, contagion), they also made another important point unique to this body of literature—that the approach which uses market data may produce systemic measures that are not directly connected to any particular theory, and that these measures could support a more efficient regulation (p. 109). Obviously, a similar point applies in the context of investigating systemic risk transmission.

Finally, similar to this study, Yang and Zhou (2013) point out that the identification of prime senders and receivers of information in the empirical framework of the financial network corresponds well to primary and secondary firms in the theoretical model of Jarrow and Yu (2001). Note also that the current application of Diebold and Yilmaz, 2014 typically does not allow for the role of exchange centers of credit risk information to be potentially systemically important, which is additionally considered in Yang and Zhou (2013).

3. Data

We use daily stock return data to investigate the financial shock transmission network among China's FIs. As noted by Huang et al. (2009) and Benoit et al. (2013), using the asset price data of FIs has three advantages: 1) ease of access; 2) price changes incorporate market anticipation, thereby foresight; 3) high frequency, reflecting up-to-date risk transmission architecture, thereby ensuring timely financial regulation and supervision.

We collect the original stock closing prices from the CSMAR database and clean the data as follows. First, we collect the daily stock closing prices of all the financial sector companies traded on China's A-share stock market. The sample period from January 1, 2008 to December 31, 2015 yields a preliminary sample of 51 FIs publicly traded in China. The sample period starts on January 1, 2008, because nearly half of the listed banks in China went public in 2007.⁶ Inclusion of more banks is important, as banks are an important source of international propagation of financial shocks (Peek and Rosengren, 1997; Imai and Takarabe, 2011; Cetorelli and

⁶ During 2007, the Industrial Bank went public in February, the China CITIC Bank in April, the Bank of Communications in May, the Bank of Nanjing and the Bank of Ningbo in July, and the Bank of Beijing and the China Construction Bank in September.

Table 1

The sample and summary statistics.

Financial institution	Abbr.	Stock code	Sector	2008–2015		2011-2015	
				Mean	Std.D	Mean	Std.D
Shaanxi International Trust	SIT	000563	Trust	0.0319	3.368	0.1005	3.178
Sinolink Securities	SLS	600109	Securities	0.0061	3.508	0.0688	3.231
Guo Yuan Securities	GYS	000728	Securities	-0.0095	3.315	0.0604	2.867
Haitong Securities	HTS	600837	Securities	-0.0143	3.286	0.0456	2.744
Pacific Securities	PS	601099	Securities	-0.0436	3.296	0.0233	2.886
Changjiang Securities	CJS	000783	Securities	-0.0047	3.360	0.0664	2.926
CITIC Securities	CS	600030	Securities	-0.0267	3.031	0.0393	2.704
Northeast Securities	NES	000686	Securities	-0.0051	3.491	0.0381	3.076
Ping An Insurance (Group) Company of China	PAI	601318	Insurance	-0.0134	2.606	0.0225	2.209
China Life Insurance Company Limited	CLI	601628	Insurance	-0.0330	2.587	0.0241	2.320
China Pacific Insurance (Group) Co., Ltd.	CPI	601601	Insurance	-0.0236	2.676	0.0209	2.316
Huaxia Bank	HXB	600015	Bank	0.0024	2.614	0.0437	2.164
Bank of China	BOC	601988	Bank	-0.0245	1.833	0.0196	1.695
Bank of Nanjing	BON	601009	Bank	0.0057	2.466	0.0515	2.158
China Merchants Bank	CMB	600036	Bank	-0.0293	2.386	0.0277	1.926
Industrial Bank	IB	601166	Bank	-0.0049	2.681	0.0459	2.236
Industrial and Commercial Bank of China	ICBC	601398	Bank	-0.0292	1.780	0.0078	1.509
Bank of Ningbo	BN	002142	Bank	-0.0036	2.580	0.0358	2.293
Ping An Bank	PAB	000001	Bank	-0.0104	2.682	0.0326	2.323
China Minsheng Bank	MSB	600016	Bank	-0.0027	2.370	0.0637	2.108
China Construction Bank	CCB	601939	Bank	-0.0242	1.961	0.0208	1.752
China CITIC Bank	CB	601998	Bank	-0.0156	2.497	0.0282	2.324
Bank of Beijing	BB	601169	Bank	-0.0167	2.436	0.0193	2.101
Bank of Communications	BC	601328	Bank	-0.0402	2.239	0.0185	1.938
Shanghai Pudong Development Bank	PDB	600000	Bank	-0.0142	2.589	0.0463	2.056
New financial institutions included during 2011-201	15						
Huatai Securities	HuaT	601688	Securities			0.0349	2.870
Guangfa Securities	GFS	000776	Securities			-0.0228	2.901
China Merchants Securities	CMS	600999	Securities			0.0311	2.775
Industrial Securities	IS	601377	Securities			0.0213	3.055
Everbright Securities	ES	601788	Securities			0.0426	2.950
Agricultural Bank of China	AB	601288	Bank			0.0167	1.587
China Everbright Bank	EB	601818	Bank			0.0055	2.017

Notes: Abbr. represents name abbreviations for financial institutions. Std.D means standard deviation. There were 1945 and 1213 observations during 2008–2015 and 2011–2015, respectively.

Goldberg, 2012; Schnabl, 2012; Kamber and Thoenissen, 2013; Alpanda and Aysun, 2014). Moreover, China's financial system has been traditionally dominated by banks, especially by the Big Four. Hence, the beginning of the sample period enables us to include a sufficient number of listed banks (14 banks, including three of the Big Four) while also facilitating an examination of the impact of the 2008 global financial crisis. As a robustness check below, we also consider an alternative sample period starting on January 1, 2011 which incorporates all the 16 currently listed banks in China (including all the Big Four).

Second, we exclude the institutions that cannot satisfy the following two conditions from the preliminary sample: 1) the stock is continuously traded during the sample period without being suspended for a substantial time period; 2) the missing observations are on average fewer than 20 trading days (one month) per year. Then, we obtain a final sample of 25 FIs (including 14 banks) between 2008 and 2015 and 32 FIs (including 16 banks) between 2011 and 2015.

Third, a few missing observations of FIs are replaced by the non-missing values of previous trading days. The stock returns are then calculated as the logarithmic change of the closing prices. As the prices of China's A-share stocks (except the ST-stocks) have been limited to \pm 10% fluctuations during each trading day since December 16, 1996, we replace the return value with 9.531 (-9.531) if it is higher (lower) than 10% (-10%). The details about FIs, their basic information, and the summary statistics for their stock returns are presented in Table 1.

Although the capital account is still under strict control, China is one of the world's largest countries in terms of international trade (ranked number one since 2013). Furthermore, the country holds the world's largest foreign exchange reserves. Trade is an important channel of international transmission of financial shocks. Hence, given strong economic linkages between China and the rest of the world, the empirical results of spillovers on FIs within China may well be biased without controlling for the influence from the global financial sector. Thus, the analysis also includes financial sectors of four major economies, that is, the US, the UK, Germany, and Japan. We obtain the daily US, UK, and German financial sector indices.⁷ However, we cannot find a similar composite

⁷ In the following robustness check, we also consider using bank indices instead of financial sector indices and the basic results remain the same.

financial sector index for Japan, as there are four Tokyo Stock Exchange indices that exist separately for banks, securities firms, insurance companies, and other financial firms in Japan. Accordingly, we conducted a principal component analysis to extract the common factors underlying these four indices. The first principal component explains approximately 84% of the variation in these four indices, which is high enough to capture the common movements in the financial sector (Yang and Zhou, 2013).⁸ We thus use it as a proxy for the financial sector index in Japan. The original data of the UK and Japanese indices are collected from the CEIC database, while the US index data are collected from the website of S&P Dow Jones Indices (http://us.spindices.com/), and the German index data are collected from Bloomberg.

4. Empirical methodology

.. .

...

We propose a modification to recently developed financial network analysis (Diebold and Yilmaz, 2014) to investigate the transmission of financial shocks among Chinese FIs. The approach is built on forecast error variance decomposition of Generalized Vector Autoregression (GVAR; Pesaran and Shin, 1998; Yang et al., 2006), which provides natural and insightful measures of connectedness to explore the weighted and directed networks (Diebold and Yilmaz, 2014). As the first step, we assume the datagenerating process of the stock returns of Chinese FIs and the financial sectors of the four major foreign countries (i.e., the US, the UK, Germany, and Japan) follow an N-dimensional covariance-stationary VAR system:

$$X_t = \sum_{i=1}^{p} \Phi_i X_{t-1} + \alpha + \varepsilon_t \tag{1}$$

where *X* is a vector of the stock (or financial market) returns, α is the deterministic component of the VAR system, and $\varepsilon \sim (0, \Sigma)$ is a vector of independently and identically distributed disturbances.

The moving average representation of Eq. (1) can then be written as $X = \sum_{i=0}^{\infty} A_i e_{t-i}$, where A_i is the $N \times N$ coefficient matrix obeying the recursive rule of $A_i = \Phi_1 A_{i-1} + \Phi_2 A_{i-2} + \dots + \Phi_p A_{i-p}$, and A_0 is an $N \times N$ identity matrix with $A_0 = 0$ for i < 0. The estimated coefficients of (1) are difficult to interpret due to overparameterized and complicated interactions among the variables. As a consequence, the moving average coefficients (or their further transformations such as impulse-response functions or variance decompositions) are the key elements for understanding the dynamics of the system. We use the equation to conduct a forecast error variance decomposition under the GVAR framework, which allows us to assess the fraction of the *H*-step-ahead error variance of forecasting X_i that is due to X_j ($i \neq j$) invariant to the order of the variables.⁹

The GVAR *H*-step-ahead error variance decomposition, \tilde{d}_{ij}^{gH} for H = 1, 2, ..., is

$$\widetilde{d}_{ij}^{gH} = \frac{\sigma_{jj}^{-1} \sum_{h=0}^{H-1} (e'A_h \sum e_j)^2}{\sum_{h=0}^{H-1} (e'A_h \sum A_h' e_j)}$$
(2)

where Σ is the variance matrix, ε , σ_{jj} is the standard deviation of the error term for the *j*-th equation and e_i is the selection vector, with the *i*-th element equal to one, and all other elements equal to zero. The sum of all the elements in each row of the variance decomposition table under the GVAR framework is not equal to one. Therefore, following Yang et al. (2006) and Diebold and Yilmaz, 2014, we normalize each entry of the variance decomposition matrix by its row sum:

$$d_{ij}^{gH} = \frac{\tilde{d}_{ij}^{gH}}{\sum_{j=1}^{N} \tilde{d}_{ij}^{gH}}$$
(3)

Then, based on such GVAR forecast error variance decomposition, the population financial shock transmission network can be fully shown in the connectedness table. The connectedness table (Table 2) demonstrates the central understanding of the various connectedness measures and their relationships. Its main upper-left N × N block contains the variance decompositions, with d_{ij}^H denoting the ij – th H – step variance decomposition component. Hence, according to Diebold and Yilmaz, 2014, we define the *pairwise directional connectedness* from j to i as:

$$C_{i \leftarrow j}^H = d_{ij}^H \tag{4}$$

Note that $C_{i \leftarrow j}^{H} \neq C_{j \leftarrow i}^{H}$, so there are $N^2 - N$ separate pairwise directional connectedness measures. Then we can define the *net pairwise directional connectedness* as:

$$C_{ij}^H = C_{j \leftarrow i}^H - C_{i \leftarrow j}^H \tag{5}$$

⁸ The KMO values, which evaluate the soundness of the principal component analysis, are all above 0.8 for the overall principal component analysis and for each of the four indices.

⁹ H is the connectedness horizon in the connectedness (will be demonstrated in detail later). Choosing such a horizon, as pointed out by Diebold and Yilmaz (2014), is important because it is related to issues of dynamic connectedness (in the fashion of spillovers) as opposed to purely contemporaneous connectedness. In this study, we choose 10 as the connectedness horizon, as it coheres with the 10-day value at risk required by the Basel Accord. Choosing other horizons around the value 10 might provide a way of "robustness checks," but the actual values of the connectedness might not remain similar with alternative Hs. See Diebold and Yilmaz (2014) for more details.

Table 2		
Connectedness	table	schematic.

	x_1	<i>x</i> ₂		x_N	From others
x_1	d_{11}^H	d_{12}^H		d_{1N}^{H}	$\Sigma_{j=1}^{N} d_{1j}^{H}, j \neq 1$
x_2	d_{21}^H	d_{22}^{H}		d_{2N}^{H}	$\Sigma_{j=1}^{N} d_{2j}^{H}, j \neq 2$
:	:	:	·.	:	:
x_N	d_{N1}^{H}	d_{N2}^{H}	•••	d_{NN}^{H}	$\Sigma_{j=1}^{N} d_{Nj}^{H}, j \neq N$
To others	$\Sigma_{i=1}^{N} d_{i1}^{H}, i \neq 1$	$\Sigma_{i=1}^{N} d_{i2}^{H}$, $i \neq 2$		$\Sigma_{i=1}^{N} d_{iN}^{H}, i \neq N$	$\frac{1}{N}\sum_{i,j=1}^{N} d_{ij}^{H}, i \neq j$

Source: Diebold and Yilmaz (2014).

• •

The total directional connectedness from others to i are defined as:

$$C_{i\leftarrow \bullet}^{H} = \sum_{j=1, i\neq j}^{N} d_{ij}^{H}$$
(6)

The total directional connectedness to others from i is:

$$C_{\mathbf{I}\leftarrow i}^{H} = \sum_{i=1,i\neq j}^{N} d_{ij}^{H}$$
(7)

Then, the net total directional connectedness is:

$$C_i^H = C_{\bullet\leftarrow i}^H - C_{i\leftarrow\bullet}^H \tag{8}$$

The total connectedness can be calculated as:

$$C^{H} = \frac{1}{N} \sum_{i,j=1,i\neq j}^{N} d_{ij}^{H}$$
(9)

According to Diebold and Yilmaz, 2014, the SIFIs in the above connectedness network can be defined as the ones with relatively high *total directional connectedness to others* and thus positive *net total directional connectedness*. Then, the time-varying connectedness can be obtained using the fixed rolling window approach. We follow Yang and Zhou (2013) to conduct further analysis for the determinants of such financial shock transmission network, which will be illustrated in detail later.

Finally, some important comments are in order on the modified approach proposed in this study. First, it should be noted that controlling for the influence from the financial sectors of the four major global economies on individual FIs in China is a significant difference between our empirical framework and the financial network approach proposed by Diebold and Yilmaz, 2014. This modification can thus be expected to improve the informational efficiency and accuracy of the VAR system. Without controlling for the influence from the financial sectors of the major global economies, as pointed by Kilian and Lütkepohl (2017), such a VAR system may suffer from an omitted-variable bias and become informationally deficient.

Second, the modified approach allows for more flexibility in recovering the structure of the financial network. As the financial network is composed of individual FIs, the starting point of the financial network analysis (e.g., Diebold and Yilmaz, 2014; Yang and Zhou, 2013) naturally focuses on the spillovers among individual FIs from the perspective of connectedness. However, unlike previous studies, our modified approach enables us to reveal the structure of financial network based on subgroups of individual FIs (however defined), rather than the information on individual FIs or the aggregate information across all FIs.

5. Full sample results and robustness checks

5.1. Baseline results

In what follows, we present the full sample results on the transmission of financial shocks among 25 FIs while controlling for the influences from the financial sectors of the four major global economies (i.e., the US, the UK, Germany, and Japan). Following Diebold and Yilmaz, 2014, we identify the institutions with higher positive net total directional connectedness and higher total directional connectedness to others in the financial shock transmission network as SIFIs. We also briefly address the total directional connectedness from others when discussing the financial shock transmission network below.

We model the stock returns of the 25 FIs and the financial sectors of the four global economies¹⁰ as a 1-lag VAR system with the optimal lag in Eq. (1) being selected by minimizing the Akaike Information Criterion.¹¹ Similar to previous studies, we calculate the full sample connectedness based on 10-step-ahead (i.e., two weeks) generalized forecast error variance decomposition. Table 3 shows

¹⁰ Following Bessler and Yang (2003), the four global financial sectors are modeled on a same calendar day basis with China. We will discuss the nonsynchronous trading problem later.

¹¹ The maximum lag allowed is set to 15 days (3 weeks).

 Table 3
 Full sample connectedness of 25 financial institutions and 4 major global financial sectors, 2008–2015.

ш																														1	6.	total	UK N
F Fr(0.1 89	0.1 89	0.1 91	0.1 91	0.1 90	0.1 91	0.1 92	0.1 91	0.4 91	0.3 92	0.3 92	0.2 92	0.1 91	0.1 92	0.3 92	0.2 92	0.2 91	0.1 92	0.1 92	0.2 92	0.2 92	0.2 91	0.2 92	0.3 92	0.2 92	2.6 37	8.8 57	6.8 61	7.2 53	3 TC	0 85.	dness,	narkei
ВM	2	2	-:	2	2	2	e.	2	Ŀ.	4.	4.	e.i	e.	e.	4.	с.	e.	2	e.	2	e.i	e.	e.	4.	e.	1	.4		.1	4	-1	mecte	lclal n
JPF	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	1	e	39	ŝ	15	- 46	tal con	s nnai
UKF	0.2	0.1	0.1	0.1	0.1	0.1	0.1	0.1	0.5	0.3	0.3	0.2	0.2	0.2	0.3	0.3	0.3	0.2	0.2	0.3	0.3	0.2	0.2	0.4	0.3	20.8	42.7	11	21.1	59	2	note <i>to</i>	the u
USF	0.1	0.1	0.1	0	0.1	0.1	0.1	0.1	0.4	0.2	0.2	0.1	0.1	0.1	0.2	0.1	0.2	0.1	0.1	0.1	0.2	0.1	0.1	0.3	0.1	63.1	17.8	10.4	13.2	45	8	ET de	resen
PDB	ĉ	2.7	2.7	ĉ	2.8	2.9	3.5	2.8	3.8	3.6	3.6	5.2	4	4.7	5.2	5.5	4.1	4.5	5.2	4.9	4.2	4.2	4.9	4.6	8	0.1	1.2	1.6	1	100	ø	and N	ole rep
D R C	2.9	2.3	2.6	2.8	2.6	2.8	2.9	2.5	4.1	4	4	4.8	5.5	4.5	4.9	4.6	5.5	4.6	4.6	ъ	5.4	ß	4.8	7.9	4.7	0.2	1	1.8	0.9	101	6	l, TO,	the tai
B B	3.1	2.6	2.7	3.1	2.7	2.9	ŝ	2.7	3.8	3.6	3.9	4.9	4.3	5.3	4.8	ъ	4.3	5.2	4.7	4.7	4.4	4.8	8	4.7	4.9	0.1	0.7	1.3	0.9	66	4	From	MF IN
9	2.8	2.5	2.4	2.8	2.6	2.6	2.9	2.6	3.3	3.6	3.4	4.4	ß	4.2	4.2	4.2	4.3	4.7	4.2	4.6	4.5	8.6	4.5	4.6	3.9	0.1	0.5	1.2	0.4	91	0	n. TC	and G
ŋ	2.6	2.3	2.5	2.7	2.7	2.6	2.9	2.6	3.8	3.9	3.7	4.3	6.1	4	4.4	4.1	6.1	4	4.1	4.5	8.3	4.6	4.3	5.1	4	0.2	0.8	1.6	0.5	95	ŝ	positio	JFF,
MDB	2.9	2.4	2.5	2.8	2.5	2.7	2.8	2.6	3.7	3.9	3.8	ഹ	4.7	4.4	ы	4.9	4.7	4.4	4.9	8.1	4.7	4.8	4.6	4.8	4.9	0.1	0.9	1.2	0.7	97	ß	lmorel	, UKF
PAB	3.4	2.9	с	3.2	2.9	3.1	3.4	3.1	4.1	3.7	3.8	4.9	4.1	4.6	4.8	5.4	4.1	4.8	7.9	ß	4.3	4.6	4.7	4.6	5.3	0.1	0.9	1.5	0.7	101	6	ance d	7. USF
PN P	3.9	3.3	3.5	3.4	3.4	3.5	3.5	3.5	3.7	3.8	3.9	4.8	4.6	5.8	4.5	4.9	4.2	7.7	4.9	4.7	4.3	5.2	5.4	4.7	4.7	0.1	0.7	1.2	0.6	105	13	r varia	спиел
) A	2.2	2	2.1	2.3	2.2	2.2	2.3	2.1	3.1	3.4	3.2	3.8	5.8	3.8	4.1	3.7	9.3	3.5	3.5	4.1	5.5	4	3.7	4.6	3.5	0.1	0.8	1.3	0.7	84	-7	st erro	, respe
-	3.2	2.8	2.9	3.3	2.9	ĉ	3.5	e	4.1	3.9	4	5.3	4.4	ß	5.3	7.6	4.5	ъ	5.6	5.3	4.5	4.8	5.3	4.8	5.8	0.1	0.9	1.5	0.8	90	14 -	foreca	eaness
	2.9	2.6	2.6	с	2.8	2.7	3.2	2.7	4.3	4	4	5.3	4.7	4.7	7.8	5.2	4.9	4.5	4.9	5.3	4.8	4.7	ß	ß	5.4	0.2	1.1	2.2	1	03 1	11	d VAR	conneci
NIOC	3.5	e	3.2	3.1	3.2	3.2	3.2	3.1	3.6	3.6	3.7	4.7	4	8	4.5	4.7	4.4	5.5	4.5	4.5	4.2	4.5	5.3	4.4	4.7	0.1	0.5	1.5	0.5	66	7	leralize	sctionat
3	2.3	7	2.1	2.3	2.2	2.2	2.4	2.3	3.2	3.4	3.2	3.8	9.3	3.5	3.9	3.6	5.8	3.8	3.5	4.1	5.5	4.6	3.7	4.7	3.4	0.1	0.6	1.2	0.5	84	-7	ad gen	tat atre
a QV	3.3	2.6	2.8	3.2	с	с	3.3	2.9	4	3.9	4	7.6	4.6	4.9	5.4	5.2	4.7	4.8	5.1	5.3	4.7	4.9	5.1	4.9	5.4	0.1	0.9	1.3	0.8	40	12 -	sp-ahe	ctively
1	3.9	3.5	3.8	4	3.6	3.6	4.1	3.7	5.9	6.1	8.3	3.7	3.6	3.6	3.8	3.6	3.6	3.6	3.6	3.7	3.8	3.6	3.8	3.8	3.5	0.2	1.1	1.8	0.9	7 10	ы	10-ste	respe
Ę	3.7	3.6	3.7	3.8	3.7	3.7	4.2	3.6	5.7	8.4	9	3.6	3.8	3.5	3.7	3.5	3.8	3.4	3.5	3.8	3.9	3.7	3.5	3.8	3.4	0.2	1	1.7	1.1	7	ы	from	o ouner iarket.
Ę	3.4	3.4	3.5	3.6	3.4	3.3	3.8	3.4	8.5	5.6	5.8	3.6	3.5	3.4	3.9	3.6	3.4	3.3	3.8	3.5	3.8	3.4	3.6	3.8	3.6	0.2	1.3	2.2	1.1	94	ĉ	ulated	<i>Iness i</i> cial m
	5.3	6.4	6.5	5.6	6.1	6.4	5.3	9.5	3.1	3.2	3.2	2.3	2.2	2.6	2.2	2.4	2.1	2.8	2.6	2.2	2.2	2.3	2.3	2.1	2.4	0.1	0.3	0.7	0.3	85 5	- 6	ss calc	nnecte. 1 finan
3	LC.	5.9	5.9	6.6	5.9	6.3	8.4	9	3.9	4.1	4	с С	2.6	°,	2.9	3.1	2.6	3.2	3.2	2.7	2.8	ი	2.9	2.7	3.3	0.1	0.4	1.3	0.4	:	۔ د	tednee	nat co. erman
	5.4	6.3	6.6	6.2	6.3	6	5.8	6.7	3.2	3.4	3.4	2.5	2.3	2.8	2.4	2.6	2.3	б	2.7	2.4	2.4	2.5	2.6	2.4	2.7	0	0.3	0.9	0.3	5 06	-1	onnec	irrectio
5	5.2	6.2	5.9	5.3	9.6	5.7	ß	5.8	e	3.1	3.1	2.3	2.1	2.6	2.2	2.2	2.1	2.6	2.3	2.1	2.3	2.3	2.2	2.1	2.3	0	0.3	0.9	0.2	82	8 1	mple c	, totat a
cIII	4.9	9	9	6	5.8	6.3	6.2	5.9	3.4	3.6	3.7	2.8	2.4	2.8	2.6	2.8	2.4	2.9	2.8	2.6	2.6	2.7	2.8	2.5	2.8	0	0.4	1	0.3	91	0	full sa	otners. cial m
010	5.5	6.7	9.4	5.8	6.2	6.3	5.3	6.6	3.2	3.3	3.4	2.3	2.1	2.8	2.2	2.4	2.1	2.8	2.5	2.2	2.3	2.3	2.3	2.2	2.4	0	0.2	0.6	0.3	86	-5	ts the	s <i>from</i> finan
3	4.3	11.2	5.6	4.8	5.5	5.1	4.5	5.4	2.6	2.7	2.6	1.8	1.7	2.2	1.8	1.9	1.6	2.3	2	1.7	1.7	1.9	1.8	1.6	1.9	0	0.1	0.7	0.1	70	-19	report	<i>teanes</i> . Japan
й П	11	4.4	4.7	4	4.7	4.4	3.8	4.6	2.7	2.8	с	2.3	1.9	2.5	2	2.3	2	2.7	2.5	2.2	2	2.2	2.3	2.1	2.2	0	0.3	0.5	0.3	71	- 18	s table	<i>connec</i> narket.
Ñ	,		S	Ś		S		S	I	1	I	В	ç	Ž	B		ñ	F	В	B	В			_	в	Ŧ	H	[r.	ΠF	-	۱ E	s: Thi	ncial n
	SIT	SILS	GY	ΗT	PS	C	S	NE	PA	G	CP	ΧН	BO	BO	UN UN	IB	ICE	BN	PA	MS	8	CB	BB	BC	PD	US	UK	JР	ЯŊ	ΤO	NE	Note	finar

the results echoing the schematic shown in Table 2. The result in Table 3 presents two novel findings concerning China's financial system: 1) the strikingly high total directional connectedness from others; 2) the high total directional connectedness to others and consequently, the high net total directional connectedness of the market-oriented commercial banks compared to the Big Four. In developed countries, business connection or borrowing–lending linkage is a major determinant of interconnectedness among FIs (e.g., Acharya et al., 2012; Acemoglu et al., 2012, 2015). Arguably, either business connection or borrowing–lending linkage strength may be enhanced in a more developed and integrated financial market. Additionally, an FI may be more influenced by other institutions with more exposure. Compared to the US financial market, the development of China's financial market lags and remains relatively underdeveloped. However, compared to the 70%–82% total directional connectedness from others of US FIs (Diebold and Yilmaz, 2014, Table 3, p. 126), the 89%–92% total directional connectedness from others of China's major FIs is noticeably higher. A plausible explanation for this phenomenon is that as China's financial system is still strongly controlled by the government, the asset prices of FIs share similar pricing factor, rather than being influenced by stronger inter-institution business connection. The pairwise inter-institution connection actually is indeed lower in China (Table 3) than the US (Diebold and Yilmaz, 2014, Table 3, p. 126). We can still obtain a higher total directional connectedness from others because we include more FIs in our sample and control for the influence from the financial sectors of the four major global economies.

Another interesting result presented in Table 3 is the more pronounced average influence of the market-oriented joint-stock commercial banks compared to the Big Four in the transmission of financial shocks. This finding extends the conventional argument regarding the role of banks as an important source of propagation of financial shocks (Peek and Rosengren, 1997; Imai and Takarabe, 2011; Cetorelli and Goldberg, 2012; Schnabl, 2012; Kamber and Thoenissen, 2013; Alpanda and Aysun, 2014). Although China's financial system is dominated by a large but under-developed banking system, especially the Big Four, the result presented here shows that market-oriented commercial banks (especially Huaxia Bank (HXB), China Merchants Bank (CMB), Industrial Bank (IB), Bank of Ningbo (BN), Ping An Bank (PAB), Bank of Communications (BC), and Shanghai Pudong Development Bank (PDB)) have much higher total directional connectedness to others on average (and thus a higher net total directional connectedness) than the Big Four in terms of financial shock transmission during the sample period. In line with Diebold and Yilmaz, 2014, such a finding would imply that these market-oriented joint-stock commercial banks might also need to receive more attention in the identification of SIFIs in China, perhaps a reflection of their more aggressive risk-taking culture. The finding is consistent with the recent evidence that the Big Four have dramatically improved their performance and have higher credit quality in their loan portfolio than market-oriented joint-stock commercial banks. This has been the case since the commencement of Chinese banking reforms in 2004, when the Big Four had major loan problems (Bailey et al., 2011; Hao et al., 2014). The result is also consistent with the finding that joint-stock banks have the highest persistence in both profit and risk (Lee and Hsieh, 2013). It further extends the evidence that joint-stock banks are the most technically efficient, while larger commercial banks, including the Big Four, are less technically efficient in generating deposits and loans (Huang et al., 2017), as such technical efficiency does not yet address the associated risk issue such as aggressive risk-taking. Anecdotal evidence and news reports indeed verify such a concern for some joint-stock banks.¹² Moreover, the three FIs in the insurance industry (i.e., PAI, CLI, and CPI) also exhibit an average influence resembling that of the market-oriented commercial banks, consistent with the well-known problem of aggressive risk-taking within the Chinese insurance industry during the sample period.

Of course, the more pronounced role of market-oriented joint-stock commercial banks and the emerging influence of nonbank FIs do not mean that Big Four are not important in terms of transmission of financial shocks. Rather, these findings reflect the new development of China's financial system. Since China's government began to solve the problem of non-performing loans (NPLs) in the state-owned banking system (especially for Big Four) during the late 1990s, China's banking system has undergone a series of market-oriented reforms. After addressing the NPL problem and subsequently receiving a substantial capital injection in the early 2000s, all the Big Four went public by 2010. In 2016, four of the top six banks in the world ranked by assets included the Big Four. Additionally, most of the market-oriented joint-stock commercial banks in the sample ranked among the top 50 in the world. Hence, the result might reflect that the Big Four were already under stricter supervision due to "too big to fail" concerns, with correspondingly limited operational risk-taking and potential spillovers of financial shocks in the financial system. As a further confirmation, according to Moody's Investors Service, during 2012–2015, risky wealth management product holdings as a fraction of total assets remained steady at approximately 2% for the Big Four, while these holdings increased between 2013 and 2015 for joint-stock commercial banks and local banks, reaching approximately 20% in 2015.

To further explore the pattern of financial shock spillover across various sub-sectors, we recalculate the connectedness among sectors as well as the financial sectors of the four major global economies. Table 4 reports the total directional connectedness to each institution (or market) from each subsector (or global financial market). For FIs in the securities sector, total directional connectedness from the trust, insurance, and banking sectors is approximately 4.4%, 11%, and 39%, respectively. For FIs in the insurance sector, the average total connectedness from the trust, securities, and banking sectors is approximately 2.8%, 23%, and 52%, respectively. For FIs in the banking sector, average total directional connectedness from the trust, securities, and insurance sectors is approximately 2.2%, 17%, and 11%, respectively, with total directional connectedness from nonbank FIs exceeding 30%. Therefore, although China's financial system remains dominated by the banking sector, nonbank FIs also exert considerable influence in the financial shock transmission network. China's financial system, especially the banking sector, also exerts considerable influence on

¹² Reuters. "Shanghai Pudong Development Bank's Chengdu Branch Fined By Regulator Due To Providing Loans Illegally." January 19, 2018. The fine was 462 million yuan or \$72 million, and the bad loan involved was 77.5 billion yuan or \$12 billion. Interestingly, Pu Dong was identified as a major sender of risk in this study before the incident was known to the public.

Table 4					
Total directional	connectedness	from	each	sector/market,	2008-2015.

	Total dire	ectional connectedr	ness from						Nonbanks	
	Trust	Securities	Insurance	Bank	USF	UKF	JPF	GMF		4GFM
SIT	11	35.6	11	42	0.1	0.2	0.2	0.1	57.6	0.6
SLS	4.4	48.7	10.5	36	0.1	0.1	0.2	0.1	63.6	0.5
GYS	4.7	45.9	11	37.6	0.1	0.1	0.1	0.1	61.6	0.4
HTS	4	43.3	11.4	41	0	0.1	0.2	0.1	58.7	0.4
PS	4.7	45.7	10.7	38.5	0.1	0.1	0.2	0.1	61.1	0.5
CJS	4.4	45.1	10.6	39.4	0.1	0.1	0.2	0.1	60.1	0.5
CS	3.8	40.5	12.1	42.8	0.1	0.1	0.3	0.1	56.4	0.6
NES	4.6	45.9	10.7	38.5	0.1	0.1	0.2	0.1	61.2	0.5
PAI	2.7	22.4	20.1	52.6	0.4	0.5	0.5	0.4	45.2	1.8
CLI	2.8	23.4	20.1	52.3	0.2	0.3	0.4	0.3	46.3	1.2
CPI	3	23.4	20.1	52.2	0.2	0.3	0.4	0.3	46.5	1.2
HXB	2.3	17	10.9	68.8	0.1	0.2	0.3	0.2	30.2	0.8
BOC	1.9	15.4	10.9	71.1	0.1	0.2	0.3	0.1	28.2	0.7
BON	2.5	18.8	10.5	67.4	0.1	0.2	0.3	0.1	31.8	0.7
CMB	2	16.3	11.4	68.8	0.2	0.3	0.4	0.3	29.7	1.2
IB	2.3	17.4	10.7	68.6	0.1	0.3	0.3	0.2	30.4	0.9
ICBC	2	15.2	10.8	70.9	0.2	0.3	0.3	0.2	28	1
BN	2.7	19.6	10.3	67	0.1	0.2	0.2	0.1	32.6	0.6
PAB	2.5	18.1	10.9	67.6	0.1	0.2	0.3	0.1	31.5	0.7
MSB	2.2	15.9	11	70.1	0.1	0.3	0.2	0.2	29.1	0.8
CCB	2	16.3	11.5	69.3	0.2	0.3	0.3	0.2	29.8	1
CB	2.2	17	10.7	69.3	0.1	0.2	0.3	0.2	29.9	0.8
BB	2.3	16.9	10.9	69.3	0.1	0.2	0.3	0.2	30.1	0.8
BC	2.1	15.6	11.4	69.4	0.3	0.4	0.4	0.3	29.1	1.4
PDB	2.2	17.8	10.5	68.6	0.1	0.3	0.3	0.2	30.5	0.9
USF	0	0.2	0.6	1.7	63.1	20.8	1	12.6	0.8	97.5
UKF	0.3	2	3.4	11.5	17.8	42.7	3.4	18.8	5.7	82.7
JPF	0.5	6.1	5.7	20.4	10.4	11	39	6.8	12.3	67.2
GMF	0.3	1.9	3.1	10	13.2	21.1	3.1	47.2	5.3	84.6

Notes: This table reports the total directional connectedness of the 25 financial institutions and 4 global financial sectors from each sector (Trust, Securities, Insurance, and Bank) or global financial market (US, UK, Japan, and Germany). USF, UKF, JPF, and GMF in the table are the abbreviations for the US financial market, UK financial market, Japanese financial market, and German financial market, respectively. Nonbanks: the nonbank financial sector. 4GFM: all four global financial sectors.

the financial sectors of the four major global economies. The total directional connectedness to the US, UK, Japanese, and German financial sectors from China's banking sector is 1.7%, 11.5%, 20.4%, and 10%, respectively, while it is 0.8%, 5.7%, 12.3%, and 5.3%, respectively, from China's nonbank FIs in aggregate. China's financial sector shows a positive net pairwise directional connectedness to three out of the four global financial sectors (i.e., UK, Japanese, and German). The total directional connectedness to China's financial sector (i.e., the 25 institutions) from the US, the UK, Japan, and Germany is 3.4%, 5.6%, 7.1%, and 4.4%, respectively, while the total directional connectedness to the US, the UK, Japan, and Germany from China is 2.5%, 17.2%, 32.7%, and 15.3%, respectively. The net pairwise directional connectedness between China and the US, the UK, Japan, and Germany is -0.9%, 11.6%, 25.6%, and 10.9%, respectively. Thus, China's financial sector exerts considerable influence on the financial sectors of the major global economies, especially the Japanese financial sector. This may be attributable to China's economic growth, strict capital controls, and its growing importance in the world economy, particularly in the regional economy.

5.2. Robustness checks

We conduct several robustness checks on the main results above. The first robustness check is to use the banking sector indices instead of financial sector indices to control for the influence from the financial sectors of the four major global economies.¹³ As discussed earlier, we focus on bank-dominated China's financial system, and banks can be both an important source of international propagation of financial shocks and an important channel for transmitting them. Accordingly, it might be important to determine whether the transmission pattern of financial shocks among China's FIs will change if we restrict the outside influence only to that from the banking sector, instead of the entire financial sector.

The second robustness check investigates the potential nonsynchronous trading problem. In line with Bessler and Yang (2003), our main previous results are based on modeling all financial market data matched on the same calendar day. Trading in the European (UK and German) and North American (the US) stock markets lags behind China's and Japan's on the same calendar day.

¹³ We still use the financial sector index for Germany, as we cannot find a readily available banking sector index.

Table 5		
Robustness	checks	

	Basic Results			Banking	Index		Nonsynd	chronous		32 Insti	tutions		Filtered returns			
	From	То	Net	From	То	Net	From	То	Net	From	То	Net	From	То	Net	
SIT	89	71	-18	89	71	-18	89	74	-15	90	65	-25	45	24	-21	
SLS	89	70	-19	89	70	-19	89	71	-18	91	70	-21	63	53	-10	
GYS	91	86	-5	91	86	-5	91	88	-3	92	90	-2	70	77	7	
HTS	91	91	0	91	91	0	91	91	0	93	103	10	70	76	6	
PS	90	82	-8	90	82	-8	90	83	-7	92	86	-6	65	61	-4	
CJS	91	90	-1	91	90	-1	91	92	1	93	95	2	70	78	8	
CS	92	97	5	92	97	5	92	99	7	94	106	12	69	72	3	
NES	91	85	-6	90	85	-5	91	86	-5	93	92	-1	69	71	2	
PAI	91	94	3	91	94	3	92	102	10	93	103	10	65	51	-14	
CLI	92	97	5	92	97	5	92	101	9	93	91	-2	66	56	-10	
CPI	92	97	5	92	97	5	92	101	9	93	99	6	65	53	-12	
HXB	92	104	12	92	104	12	92	105	13	94	109	15	80	95	15	
BOC	91	84	-7	91	84	-7	91	86	-5	91	76	-15	72	61	-11	
BON	92	99	7	92	99	7	92	101	9	93	101	8	75	70	-5	
CMB	92	103	11	92	103	11	92	107	15	93	99	6	80	95	15	
IB	92	106	14	92	106	14	93	108	15	94	108	14	80	97	17	
ICBC	91	84	-7	91	84	-7	91	87	-4	91	79	-12	73	64	-9	
BN	92	105	13	92	105	13	92	106	14	94	106	12	76	75	-1	
PAB	92	101	9	92	101	9	92	103	11	93	102	9	77	80	3	
MSB	92	97	5	92	97	5	92	100	8	92	89	-3	79	87	8	
CCB	92	95	3	92	95	3	92	99	7	92	88	-4	76	76	0	
CB	91	91	0	91	91	0	92	94	2	92	81	-11	74	68	-6	
BB	92	99	7	92	99	7	92	101	9	93	95	2	78	82	4	
BC	92	101	9	92	101	9	92	106	14	93	99	6	79	90	11	
PDB	92	100	8	92	100	8	92	103	11	93	100	7	79	87	8	
USF	37	45	8	37	42	5	52	35	-17	51	46	-5	36	52	16	
UKF	57	59	2	52	51	-1	68	40	-28	61	59	-2	51	64	13	
JPF	61	15	- 46	53	13	- 40	63	37	-26	57	11	- 46	45	11	- 34	
GMF	53	43	-10	49	40	-9	63	27	- 36	51	39	-12	47	47	0	
HuaT										94	107	13				
GFS										93	103	10				
CMS										93	104	11				
IS										93	96	3				
ES										93	101	8				
AB										92	88	-4				
EB										93	102	9				
TC	85.9			85.3			87.3			88.6			68.1			

Notes: This table reports the total directional connectedness from other institutions (From), to others (To), and net total directional connectedness (Net) of each financial institution in the risk transmission network obtained from four robust checks: 1) using the bank index rather than the financial sector index to control for the influence from four global financial sectors (Banking Index); 2) modeling the US, UK, and German financial sectors as the leading markets (Nonsynchronous); 3) incorporating more financial institutions (32 institutions); and 4) using the financial institutions' stock returns filtered by the returns of Shanghai A-Share Stock Index (Filtered returns). The results estimated previously (Basic Results) are also presented to facilitate comparison. TC: total connectedness.

Combining this fact with the GVAR forecast error variance decomposition, this implies that the stock markets of Japan and China are the leading markets. Therefore, following Bessler and Yang (2003), we model the US, UK, and German markets as the leading markets in the VAR system as our second robustness check.

The third robustness check is to incorporate more FIS in our sample. During 2008–2015, a number of FIs went public in China, including the last of the Big Four—the Agricultural Bank of China. We thus redefine the sample period starting from 2011 rather than 2008 to incorporate seven extra institutions in the sample, which results in 32 FIs during 2011–2015, including all the Big Four.

The fourth robustness check is to examine whether our basic results are mainly driven by the impact of common components or common factors to Chinese FIs, although macroeconomic factors may play a role (as shown below). One might argue that the high detected connectedness among China's FIs may well be caused by common trends of the stock market prices as a proxy for overall expectations of fundamentals, or common factors that drives the stock prices rather than truly reflect interconnectedness. To address this issue, we filter out all of the FIs' stock returns by regressing each return series on the return of the Shanghai A-Share Stock Index and then recast our analysis using the filtered returns.

Table 5 reports the total directional connectedness from others, to others, and the net total directional connectedness of each FI (or sector) extracted from the four robustness checks, along with the results estimated previously to facilitate comparison. Clearly, the main results remain almost the same for the 25 FIs using banking indices. The results of modeling the US, UK, and German financial sectors as the leading markets also yield very similar inferences to what we obtained based on matching the same calendar day. However, these results do show somewhat higher total directional connectedness from others, lower total directional connectedness

to others, and thus lower net total directional connectedness (highly negative) for the financial sectors of the four major global economies. Hence, if we model the analysis on this alternative definition of a trading day, China's financial sector would exhibit even higher influence on global financial sectors.¹⁴

The result derived from the alternative sample including 32 FIs during 2011–2015 is also very close to our main previous results. Similar to the other three, the Agricultural Bank of China, the latest of the Big Four to go public, also has lower total directional connectedness to others, and thus, a lower net total directional connectedness in the financial shock transmission network. Moreover, the total directional connectedness from all other FIs is somewhat higher than during 2008–2015. The total directional connectedness to the financial sectors of the four major global economies from China also increases substantially (Appendix Tables A-2 and A-3). More interestingly, the other nonbank FIs, especially several institutions in the securities sector, emerge to manifest considerable influence in the financial shock transmission network. These results again reflect the recent development of China's financial market and call for a more in-depth investigation of the dynamic transmission network of financial shocks below.

The result driven from using FIs' filtered returns shows a mitigated and yet still highly interconnected pattern of Chinese FIs, without changing our main findings. In particular, banks still dominate China's financial sector, and nonbank FIs also bear considerable influence. The market-oriented large commercial banks also still generally play a more pronounced role than the Big Four in transmitting financial shocks. Finally, although the connectedness apparently decrease as we filter out common components of FIs' returns, the results still show positive net pairwise directional connectedness from China's financial sector to the UK, Japan, and German financial sectors.¹⁵ Thus, the basic results above largely reflect true interconnectedness rather than being mainly driven by common components. In summary, the main results generally remain robust.

6. Dynamic connectedness and its determinants

In what follows, we attempt to answer two important questions arising from our previous analysis: 1) Is the transmission of financial shocks among China's FIs variable over time? 2) If so, what major factors contribute to this variation?

6.1. Dynamic connectedness

The full sample result is informative about what occurred on average during the full sample period. However, it is less helpful to ensure effective financial regulation and supervision, which requires up-to-date information about the dynamic transmission of financial shocks as well as the potential role each FI plays in the network. To this end, similar to Diebold and Yilmaz, 2014, we use a 120-trading-day (one-half year)¹⁶ fixed rolling window to extract the dynamic connectedness of each FI (or market). Figs. 1, 2, and 3 depict the extracted net total directional connectedness, the total directional connectedness from others, and the total directional connectedness to others for each FI (or market), respectively. These dynamic connectedness patterns re-confirm our previous full sample conclusions while also having several implications for financial regulation and supervision. First, both the total directional connectedness to others (Fig. 2) and the total directional connectedness to others (Fig. 3) change over time, which results in time-varying net total directional connectedness (Fig. 1). These findings imply that the role each institution plays in financial shock transmission also changes over time. The conventional approach of identifying SIFIs based on low-frequency financial indices (e.g., IMF, BIS, and FSB, 2009; Allahrakha et al., 2015; Glasserman and Loudis, 2015) may fail to capture these dynamic changes. Thus, these indices are hardly able to serve the full purpose of efficient regulation and supervision.

Second, the dynamic financial shock transmission patterns confirm the dominant role of banks in China's financial system as well as the growing importance of nonbanking FIs. Consistent with the full sample result, the dynamic net total directional connectedness of banks exhibits a higher frequency of positive net influence in the financial shock transmission network (Fig. 1-Panel B) than the nonbank FIs (Fig. 1-Panel A). However, the 11 nonbank FIs (especially the three insurance sector FIs and a few institutions in the securities sector) each exert a positive net influence in the financial shock transmission network during most of the sample period.

Third, the market-oriented commercial banks have much greater influence than the Big Four in the financial shock transmission network. The SIFIs identified from the previous full sample connectedness table, that is, Huaxia Bank (HXB), China Merchants Bank (CMB), and Industrial Bank (IB) among others, exert a positive net influence more frequently in the financial shock transmission network. Nevertheless, an interesting result of the three incorporated Big Four banks is that they exert a positive net influence during the turmoil period (2008 financial crisis and the 2015 Chinese stock market crash) but a negative net influence during other tranquil periods in the financial shock transmission network.

6.2. Determinants of dynamic connectedness

We have investigated the full sample and the dynamic transmission of financial shocks among China's FIs. A natural question is then, what are the major factors that produced such a network? To answer this question, we follow Yang and Zhou (2013) and

¹⁴ See Appendix Table A-1 for the full sample connectedness table of modeling the US, UK, and German financial sectors as the leading markets in the alternative definition of a trading day.

¹⁵ See Appendix Table A-4 for the full sample connectedness table of using the filtered returns.

¹⁶ In China, trading days of the stock market are roughly 240 days per year, due to additional public holidays such as the 1-week Spring Festival, and so on.



Fig. 1. Dynamic net total directional connectedness, 2008–2015. A: 11 nonbank financial institutions and 4 global financial sectors B: 14 banks

conduct further analysis. Before conducting the analysis, we use a 120-trading-day fixed rolling window to extract the total directional connectedness from others, to others, and the net total directional connectedness of each FI using the expanded sample of 32 FIs during 2011–2015. Incorporating more FIs will help facilitate our investigation of the firm-specific determinants. To serve as a further robustness check, as shown in Fig. 4, we verify that the dynamics of net total directional connectedness of the 25 FIs during 2008–2015 and during 2011–2015 are strongly similar, thus confirming again the robustness of the main results above.

Table 6 reports the summary statistics of dynamic total directional connectedness from others, to others, and the net total directional connectedness of the 32 FIs. Again, these summary statistics confirm our previous conclusions: 1) Banks play a central role in the transmission of financial shocks; 2) Nonbank FIs also have a considerable influence in the financial shock transmission network; and 3) Market-oriented commercial banks typically play a more pronounced role than the Big Four in financial shock



transmission network. In the following analysis, we will use the connectedness measured at the end of a month (or quarter) to explore how various factors at monthly (or quarterly) intervals could affect the spillover pattern.

6.2.1. Macroeconomic factors

In this section, we will investigate whether the transmission of financial shocks is influenced by macroeconomic factors. The impact of macroeconomic factors on the performance and risk of FIs can be even more pronounced than firm-specific factors, as suggested by Collin-Dufresne et al. (2001). We will comprehensively examine a number of macroeconomic factors in



Fig. 2. Dynamic total directional connectedness from others, 2008–2015. A: 11 nonbank financial institutions and 4 global financial sectors B: 14 banks



China.¹⁷ Following Yang and Zhou (2013), as a preliminary analysis, we will first use a simple regression based on Newey-West robust standard errors to examine whether a certain macroeconomic factor (or various indicators of the same factor) impacts the connectedness (net, from, to) of an FI with both statistical and (at least some) economic significance.¹⁸ Then, based on the results of

¹⁷ All macroeconomic factor variables are collected form the CEIC database.

¹⁸As a very preliminary prescreening measure, we consider variables with explanatory power equal to or > 1% (i.e., with adjusted- R^2 equal to or > 1%) as meeting the minimum threshold of economic significance.



Fig. 3. Dynamic total directional connectedness to others, 2008–2015. A: 11 nonbank financial institutions and 4 global financial sectors B: 14 banks

these simple regressions, we will further conduct multiple regressions based on Newey-West robust standard errors to finalize the selection of comparatively important factors, after controlling for collinearity of these factors.

First, we examine whether the transmission of financial shocks is affected by various monetary policy measures. An important monetary measure in China is the money supply, particularly M2 and its various components (quasi money and its three components, i.e., saving deposits, time deposits, and other deposits). As reported in Table 7-A, only quasi money and its component of other deposits impact total directional connectedness from others with both statistical and economic significance (Panel A of Table 7-A). This finding implies that financial shock transmission is affected by M2, mainly through changes in the 'other deposit'.



component.

Second, with interest rate liberalization, short-term interest rates are also increasingly becoming the monetary policy target in China. We thus examine the role of different money market interest rates, that is, the Shanghai Interbank Offered Rates (SHIBOR), with maturities from overnight to one year. We find that longer-maturity SHIBOR (6-month, 9-month, and 1-year) affect total



Fig. 4. Net total directional connectedness of 25 financial institutions extracted from 2008 to 2015 and 2011–2015 samples. Notes: The black line and blue line represent the net total directional connectedness estimated from the 2008–2015 25-financial-institution sample and the 2011–2015 32-financial-institution sample, respectively.

directional connectedness from others with both statistical and economic significance (Panel B of Table 7-A), with longer-maturity SHIBOR having an even more significant impact on financial shock transmission.

Third, we examine whether the transmission of financial shocks is affected by the development of informal financial systems or the shadow banking system in China. These systems have been suggested as potentially destructive factors in China's financial system (e.g., Allen et al., 2012). Due to poor data availability, we used a very limited proxy to examine the three most popular informal finance measures in China. These included emerging Internet finance as measured by Yu'e Bao¹⁹ 7-day annualized return as the proxy, the informal credit market measured by the Wenzhou private lending rate (average, automobile, and real estate mortgage), and the shadow banking system as proxied by deposit and portfolio investments of the insurance sector. We found that the Yu'e Bao 7-day annualized return and the Wenzhou private lending rate (especially for automobile mortgage lending) had both statistically and economically significant impact on the total directional connectedness from others for the transmission of financial shocks (Panel C of Table 7-A).

Fourth, we examine whether the transmission network of financial shocks is affected by the RMB exchange rate with increasing internationalization, as measured by currency swap programs between China and other countries. Years of continuing RMB appreciation and rapid increases in China's foreign exchange reserves suggest there is a large amount of speculative "hot" money, which is a potentially destructive force in China's financial system (Allen et al., 2012). Specifically, we examine the influence of China's real effective exchange rate and different terms of currency swap rates (i.e., one week, one month, three months, six months, nine months, and one year), finding that the real effective exchange rate has both statistically and economically significant impact on the total directional connectedness from others in the transmission network of financial shocks (Panel D of Table 7-A).

Fifth, we examine whether the transmission network of financial shocks is affected by the various Banking Climate Indices (BCIs)

¹⁹ Yu'e Bao is sponsored and managed by Alibaba, the largest Internet commercial company in China, and became the world's largest money market fund in 2017.

Table 6

Summary statistics of estimated dynamic connectedness 2011-2015.

	From				То				Net			
	Mean	Std.D	Min	Max	Mean	Std.D	Min	Max	Mean	Std.D	Min	Max
SIT	0.901	0.035	0.728	0.938	0.713	0.193	0.241	1.037	-0.188	0.164	-0.537	0.121
SLS	0.904	0.055	0.689	0.95	0.798	0.2	0.272	1.118	-0.106	0.155	-0.505	0.172
GYS	0.932	0.015	0.887	0.954	0.998	0.137	0.701	1.214	0.067	0.125	-0.209	0.264
HTS	0.939	0.008	0.892	0.953	1.096	0.133	0.555	1.373	0.157	0.131	-0.381	0.441
PS	0.908	0.059	0.696	0.949	0.872	0.193	0.371	1.124	-0.037	0.143	-0.408	0.178
CJS	0.931	0.016	0.872	0.952	0.989	0.107	0.621	1.198	0.058	0.097	-0.266	0.278
CS	0.94	0.008	0.919	0.953	1.108	0.092	0.803	1.289	0.168	0.092	-0.135	0.35
NES	0.927	0.014	0.875	0.946	0.925	0.11	0.669	1.294	-0.002	0.104	-0.251	0.356
HUAT	0.939	0.009	0.911	0.952	1.103	0.092	0.942	1.359	0.164	0.091	0.001	0.425
GFS	0.937	0.01	0.905	0.951	1.056	0.124	0.813	1.364	0.12	0.121	-0.117	0.426
CMS	0.935	0.011	0.898	0.951	1.055	0.124	0.716	1.371	0.119	0.12	-0.182	0.442
IS	0.933	0.013	0.898	0.949	1.015	0.116	0.683	1.198	0.082	0.109	-0.224	0.271
ES	0.934	0.01	0.901	0.95	1.023	0.103	0.753	1.277	0.089	0.1	-0.165	0.348
PAI	0.934	0.01	0.903	0.947	1.017	0.077	0.875	1.253	0.083	0.075	-0.058	0.322
CLI	0.929	0.012	0.892	0.95	0.961	0.135	0.677	1.244	0.032	0.129	-0.226	0.314
CPI	0.932	0.011	0.899	0.949	0.969	0.076	0.824	1.183	0.038	0.071	-0.105	0.239
Nonbank av.	0.928	0.019	0.860	0.950	0.981	0.126	0.657	1.244	0.053	0.114	-0.236	0.309
Bank av.	0.927	0.020	0.858	0.951	0.951	0.131	0.628	1.212	0.024	0.116	-0.246	0.273
HXB	0.936	0.012	0.902	0.952	1.039	0.079	0.818	1.208	0.103	0.072	-0.092	0.278
BOC	0.905	0.042	0.753	0.947	0.761	0.186	0.329	1.16	-0.144	0.149	-0.443	0.22
BON	0.936	0.012	0.893	0.952	1.06	0.1	0.734	1.242	0.124	0.095	-0.198	0.298
CMB	0.936	0.01	0.908	0.952	1.035	0.103	0.748	1.27	0.098	0.101	-0.176	0.34
IB	0.939	0.009	0.914	0.953	1.086	0.089	0.826	1.283	0.147	0.087	-0.093	0.342
ICBC	0.905	0.033	0.792	0.948	0.723	0.19	0.364	1.145	-0.182	0.162	-0.444	0.198
BN	0.938	0.011	0.908	0.955	1.065	0.081	0.894	1.264	0.127	0.076	-0.042	0.338
PAB	0.933	0.016	0.875	0.953	0.999	0.12	0.718	1.241	0.067	0.109	-0.159	0.303
MSB	0.93	0.016	0.873	0.949	0.945	0.104	0.657	1.136	0.015	0.091	-0.217	0.202
CCB	0.91	0.03	0.798	0.95	0.785	0.18	0.435	1.168	-0.125	0.155	-0.428	0.219
CB	0.921	0.032	0.807	0.951	0.895	0.2	0.424	1.177	-0.026	0.172	-0.393	0.228
BB	0.931	0.019	0.868	0.952	0.983	0.14	0.623	1.248	0.053	0.125	-0.246	0.303
BC	0.931	0.015	0.877	0.952	0.962	0.131	0.666	1.212	0.031	0.121	-0.217	0.284
PDB	0.936	0.015	0.885	0.953	1.052	0.126	0.75	1.28	0.116	0.114	-0.139	0.342
AB	0.912	0.034	0.799	0.947	0.801	0.163	0.392	1.155	-0.111	0.134	-0.418	0.213
EB	0.935	0.016	0.879	0.951	1.03	0.106	0.666	1.2	0.095	0.095	-0.23	0.255

Notes: This table reports the summary statistics of estimated dynamic total directional connectedness from others (From), to others (To), and net total directional connectedness (Net) of 32 financial institutions using a 120-trading-day fixed rolling window. Std.D: standard deviation; av.: average.

constructed by the PBOC. BCIs involve a wide range of macroeconomic activities that are closely related to the operation of banks. BCIs also serve as an important reference for financial regulation and supervision in China. We examine the BCIs for the degree of economic overheating, the industry climate, bankers' confidence, money policy sentiment, profitability, demand for various loans (manufacturing and nonmanufacturing, large, medium, and small and micro enterprises, and so on), and loan approvals. Our findings reveal that only the BCI of money policy sentiment impacts the total directional connectedness from others in the transmission network of financial shocks with both statistical and economic significance (Panel E of Table 7-A).

Sixth, we explore the role of the Chinese real estate market, especially the funding sources of real estate investment. China's booming real estate market, especially its soaring housing prices, has attracted worldwide attention during the past decade. For example, the IMF (2011) lists "potential steep price correction in Chinese property markets" as a major risk to global recovery from the financial crisis. Allen et al. (2012) noted the potentially destructive outcomes for China's financial system if turmoil emerges in the Chinese real estate market. Nevertheless, our findings do not show that real estate market investment affects the transmission network of financial shocks with either statistical or economic significance. Almost all types of real estate investment funding sources (domestic loans, foreign direct investment, self-raised funding, and other funds such as deposits, advanced payment, and mortgage) affect the total directional connectedness from others in the transmission network of financial shocks with statistical but not economic significance (Panel F of Table 7-A).

Lastly, we also examine whether the transmission network of financial shocks is affected by the fiscal budget/surplus, revenue, and expenditures of the central government and local governments, respectively. Since the 1994 tax reform in China, a large portion of local government revenue must be reallocated by the central government. This reallocation induces local governments to depend heavily on land transformation and the so-called financial platform firms to finance their public expenditures. Such local government behaviors are deemed as the hands pushing China's booming real estate market, with potentially destructive effects for China's

Table 7

Macroeconomic factors.

7-A: Simple regressions

	Net			From			То			
	Estimate	Obs.	adj-R ²	Estimate	Obs.	adj-R ²	Estimate	Obs.	adj-R ²	
Panel A: Money policy—money supply, M2, year to year grow	rth (%)									
Money supply-M2	0.128	1728	-0.000	-0.046	1728	-0.000	0.082	1728	-0.00	
	(0.446)			(0.065)			(0.492)			
M2: Quasi money	-0.132	1728	-0.000	-0.240^{***}	1728	0.030	-0.372	1728	0.001	
	(0.304)			(0.037)			(0.334)			
M2, Quasi money: saving deposit	-0.078	1728	-0.000	-0.029	1728	0.001	-0.106	1728	-0.0	
	(0.203)			(0.031)			(0.227)			
M2, Quasi money: time deposit,	-0.019	1728	-0.001	-0.009	1728	-0.000	-0.027	1728	-0.0	
	(0.145)			(0.019)			(0.160)			
M2, Quasi money: other deposit	0.004	1728	-0.001	-0.019***	1728	0.026	-0.015	1728	-0.0	
	(0.036)			(0.006)			(0.040)			
and B. Charachai Intertents Officer J Bater (CUUDOD) (0/)										
anel B: Snangnal Interbank Offered Rates (SHIBOR) (%)	0.010	1700	0.001	0.160+*	1700	0.000	0.144	1700	0.0	
Jvernight	-0.018	1728	-0.001	0.162**	1728	0.003	0.144	1728	-0.0	
	(0.563)			(0.075)			(0.619)			
week	-0.091	1728	-0.001	0.070	1728	0.000	-0.021	1728	-0.0	
	(0.577)			(0.073)			(0.635)			
week	-0.106	1728	-0.001	0.059	1728	0.000	-0.046	1728	-0.0	
	(0.542)			(0.072)			(0.600)			
month	-0.098	1728	-0.001	0.078	1728	0.000	-0.021	1728	-0.0	
	(0.578)			(0.074)			(0.639)			
month	-0.389	1728	-0.000	-0.240^{*}	1728	0.005	-0.629	1728	0.000	
	(0.810)			(0.125)			(0.906)			
month	-0.684	1728	0.000	-0.540***	1728	0.018	-1.224	1728	0.002	
	(1.036)			(0.153)			(1.147)			
month	-0.749	1728	0.000	-0.720^{***}	1728	0.027	-1.469	1728	0.002	
	(1.123)			(0.163)			(1.242)			
year	-0.764	1728	0.000	-0.779***	1728	0.029	-1.543	1728	0.002	
	(1.175)			(0.163)			(1.295)			
anel C: Informal financial sectors: Internet finance, folk credi	it market, shad	low bank	ung system							
u'e Bao 7-day anualized return (%)	-0.333	992	-0.001	-0.522***	992	0.023	-0.855	992	0.001	
	(1.089)			(0.183)			(1.238)			
Venzhou private lending rate: average (%)	-0.215	1376	-0.000	-0.256***	1376	0.020	-0.471	1376	0.001	
	(0.443)			(0.054)			(0.489)			
Venzhou private lending rate: automobile mortgage (%)	-0.137	1376	-0.001	-0.394**	1376	0.018	-0.531	1376	0.000	
	(0.882)			(0.154)			(1.005)			
Venzhou private lending rate: real estate mortgage (%)	-0.185	1376	-0.000	-0.095*	1376	0.001	-0.281	1376	-0.0	
	(0.419)			(0.051)			(0.460)			
nsurance sector: bank deposit, RMB bn (log-)	3.366	1728	0.000	-0.502	1728	0.000	2.864	1728	-0.0	
	(5.650)			(0.772)			(6.257)			
nsurance sector: portfolio investment, RMB bn (log-)	1.395	1728	-0.000	0.361	1728	0.001	1.756	1728	0.000	
	(2.736)			(0.364)			(3.010)			
and D. Eveloped acts and summer and										
anei D: Exchange rate and currency swaps	0.047	1500	0.000	0.040	1800	0.001	0.005	1700	0.000	
eal efficient exchange rate $(2010 = 100)$	0.047	1728	0.000	0.048***	1728	0.024	0.095	1728	0.002	
	(0.079)			(0.009)			(0.086)			
urrency swap: 1 week (%)	-0.000	1728	-0.001	-0.001**	1728	0.000	-0.001	1728	-0.0	
	(0.003)			(0.000)			(0.003)			
urrency swap: 1 month (%)	-0.000	1728	-0.001	0.002***	1728	0.002	0.002	1728	-0.0	
	(0.009)			(0.001)			(0.010)			
urrency swap: 3 month (%)	-0.025	1728	-0.000	-0.018***	1728	0.004	-0.043	1728	0.00	
	(0.035)			(0.006)			(0.040)			
urrency swap: 6 month (%)	-0.012	1728	-0.000	-0.014^{***}	1728	0.007	-0.027	1728	0.000	
	(0.029)			(0.005)			(0.033)			
urrency swap: 9 month (%)	-0.027	1728	-0.000	-0.028***	1728	0.009	-0.055	1728	0.000	
	(0.055)			(0.010)			(0.063)			
Currency swap: 1 year (%)	-0.017	1728	-0.000	-0.019**	1728	0.005	-0.035	1728	-0.0	
S TE ST ST	(0.041)	. =0		(0.007)	. ===		(0.047)	. ===		
	((/)			(,)			
anel E: Banking Climate Indices (BCI, %)										
CI: degree of economy was overheated	-0.025	1728	-0.000	0.012	1728	0.002	-0.012	1728	-0.0	
	(0.068)			(0.009)			(0.075)			
CI: industry climate	-0.046	1728	-0.001	-0.011	1728	0.000	-0.057	1728	-0.0	
	(0.095)			(0.011)			(0.103)			

(continued on next page)

Table 7 (continued)

7-A: Simple regressions

	Net			From			То		
	Estimate	Obs.	adj-R ²	Estimate	Obs.	adj-R ²	Estimate	Obs.	adj-R ²
BCI: bankers' condifence	-0.002	1728	-0.001	0.003	1728	-0.000	0.001	1728	-0.001
BCI: money policy sentiment	0.008	1728	-0.001	0.059 ***	1728	0.047	0.067	1728	0.001
BCI: Profitability	(0.071) -0.041	1728	-0.000	-0.021**	1728	0.003	(0.079) -0.062	1728	0.000
BCI: loan demand	(0.034) -0.039	1728	-0.000	-0.001	1728	-0.001	(0.091) -0.040 (0.100)	1728	-0.000
BCI: loan demand, manufacturing	(0.091) - 0.047	1728	-0.000	-0.002	1728	-0.001	(0.100) -0.048 (0.102)	1728	-0.000
BCI: loan demand, non-manufacturing	(0.093) -0.062 (0.164)	1728	-0.000	0.000	1728	-0.001	(0.102) -0.062 (0.180)	1728	-0.000
BCI: loan demand, large enterprise	(0.104) -0.119 (0.209)	1728	0.000	-0.006	1728	-0.001	(0.130) -0.126 (0.230)	1728	0.000
BCI: loan demand, medium enterprise	(0.209) -0.049 (0.100)	1728	-0.000	(0.028) -0.003 (0.014)	1728	-0.001	(0.230) -0.052 (0.110)	1728	-0.000
BCI: loan demand, small and micro enterprise	(0.109) -0.045	1728	-0.000	(0.014) -0.012	1728	0.001	(0.119) -0.057	1728	-0.000
BCI: loan approval	(0.093) 0.100 (0.230)	1728	-0.000	(0.011) 0.055* (0.032)	1728	0.005	(0.101) 0.156 (0.256)	1728	0.000
Panel F: Real Estate market: real estate climate and real estate in	vestment (F	EI)		(0.002)			(01200)		
Real estate climate (index, 2000 = 100)	- 0.100 (0.323)	1728	-0.000	0.052 (0.043)	1728	0.001	-0.048 (0.357)	1728	-0.001
REI: domestic loans, RMB mn (log-)	0.313	1728	-0.000	-0.516^{***}	1728	0.009	-0.203 (1.202)	1728	-0.001
REI: foreign invesment, RMB mn (log-)	-0.050 (0.832)	1728	-0.001	-0.145	1728	0.001	-0.194	1728	-0.001
REI: foreign invesment, direct invesmtne, RMB mn (log-)	-0.040 (0.833)	1728	-0.001	-0.154 (0.155)	1728	0.001	-0.194	1728	-0.001
REI: self raised, RMB mn (log-)	0.186	1728	-0.001	-0.391***	1728	0.007	-0.204	1728	-0.001
REI: self raised, self owned, RMB mn (log-)	0.100	1728	-0.001	-0.414^{***} (0.146)	1728	0.008	-0.314	1728	-0.001
REI: other funds, RMB mn (log-)	0.262	1728	-0.000	-0.276^{**} (0.126)	1728	0.004	-0.014	1728	-0.001
REI: other funds, deposits & advanced payment, RMB mn (log-)	0.260	1728	-0.000	-0.292^{**} (0.124)	1728	0.005	-0.033	1728	-0.001
REI: other funds, mortgage, RMB mn (log-)	0.326 (0.810)	1728	-0.000	- 0.242* (0.124)	1728	0.003	0.084 (0.898)	1728	-0.001
Panel G: Government surplus, revenue, and expenditure (RMB bn	ı, log-)								
Government surplus	0.073 (0.712)	1728	-0.001	-0.266** (0.120)	1728	0.001	-0.193 (0.793)	1728	-0.001
Government revenue	1.420	1728	-0.000	0.383	1728	0.000	1.803 (2.604)	1728	-0.000
Government expenditure	0.543	1728	-0.000	0.545***	1728	0.004	1.088	1728	-0.000
Central government surplus	-0.015	1664	-0.001	-0.392^{***}	1664	0.003	-0.406	1664	-0.001
Central government revenue	0.688	1664	-0.000	-0.208	1664	-0.000	0.481	1664	-0.001
Central government expenditure	(1.325) 1.086 (1.010)	1664	-0.000	(0.213) 0.665**	1664	0.003	(1.4/9)	1664	-0.000
Local government surplus	0.068	1664	-0.001	(0.262) - 0.358**	1664	0.001	(2.098) -0.290	1664	-0.001
Local government revenue	(0.979) 1.502	1664	-0.000	(0.182) 0.736*	1664	0.003	(1.105) 2.238	1664	0.000
Local government expenditure	(2.486) 0.611	1664	-0.000	(0.388) 0.659***	1664	0.007	(2.776) 1.270	1664	0.000

(continued on next page)

Table 7 (continued)

Money supply, N

7-B: Multiple regressions

gressions			
	Dependent variable	s	
	(1)	(2)	(3)
	Net	From	То
A2, Quasi money: other deposit, YoY-growth	-0.020	-0.060***	-0.080
	(0.062)	(0.013)	(0.073)
	-0.137	-0.761***	-0.898

	(2.078)	(0.243)	(2.296)
Yu'e Bao 7-day anualized return	-0.386	0.028	-0.358
	(1.929)	(0.360)	(2.229)
Wenzhou private lending rate: automobile mortgage	0.176	0.092	0.269
	(0.863)	(0.163)	(0.987)
Real efficient exchange rate	-0.026	0.189***	0.163
	(0.193)	(0.036)	(0.222)
Banking climate indices: money policy sentiment	0.022	-0.032	-0.011
	(0.144)	(0.020)	(0.161)
Constant	6.701	75.508***	82.209***
	(25.835)	(4.157)	(29.175)
Observations	992	992	992
Adj-R ²	-0.005	0.239	0.003

Notes: This table reports the results of multiple regressions with robust standard errors. The heteroscedasticity and autocorrelation consistent standard errors (HAC) are in parentheses. "*", "**", and "***" denote significance at 10%, 5%, and 1%, respectively.

financial system (Allen et al., 2012). Nevertheless, our study results do not reveal that any of these related variables are economically significant (Panel G of Table 7-A).

Based on the above results of simple regressions, we further perform a set of multiple variable robust regressions to determine the comparatively important factors, as many of these factors may be related to one another. Table 7-A reports the results, showing that only monetary policy-related factors (i.e., other deposits of quasi money in M2 [negative], 1-year SHIBOR [negative], and real effective exchange rates [positive]) have both statistically and economically significant explanatory power for the total directional connectedness from others. However, these factors have neither statistical nor economic significant influence on either total directional connectedness to others or the net total directional connectedness. Hence, there are macroeconomic factors, especially monetary policy measures, which determine the degree of influence by others in the transmission network of financial shocks.

6.2.2. Firm-specific determinants

In the following sections, we will investigate whether and how the transmission of financial shocks in China is influenced by firmspecific factors, as commonly discussed in the literature (e.g., Chen et al., 2010; Li et al., 2019). First, we examine the influence of leverage ratios (i.e., total debt to total assets, long-term debt to total assets, and short-term debt to total assets). Yang and Zhou (2013) also find that the short-term debt ratio is one of the significant determinants affecting credit risk spillovers among American and European banks around the time of the recent global financial crisis. We find that the long-term debt to total assets ratio positively influences the total directional connectedness from others, to others, and the net total directional connectedness, while the short-term debt to total assets ratio negatively influences these types of connectedness. The net result is that the total debt to total assets ratios loses statistical significance.²⁰

Second, we examine whether the transmission of financial shocks is affected by FIs' (short-term) liquidity, as measured by the accounts receivable turnover, the ratio of liquid assets to total assets, the ratio of current assets to total assets, and receivables to total asset. Accounts receivable turnover is an important proxy for short-term liquidity, and has a negative influence, which decreases the influence of an FI in the shock transmission networks. Similarly, we find that the ratio of liquid assets negatively affects the influence of an FI in the financial shock transmission network (net, from, and to), while the current assets ratio and receivable assets generally have no statistical significance. These findings are also generally consistent with the above negative influence of short-term debt to total assets.

Third, we consider the influence of FI size. Largely consistent with the insignificant results of Yang and Zhou (2013) concerning the largest American and European FIs, we find the size may even have a negative impact on the transmission of financial shocks among Chinese FIs. This finding may add another caveat on the conventional argument of "too big to fail," with additional evidence from China. Nevertheless, as most of the FIs studied are among the largest in China, as noted by Yang and Zhou (2013), the result does not necessarily mean that their size does not affect the roles of risk transfer. Rather, the result only implies that among the largest FIs, the relative size may not be related to the relative influence within the financial shock transmission network. On the contrary, there is an alternative explanation. Size is one of the key factors suggested by the Basel Committee for the identification of

²⁰ According to the definitions, total debt to total asset = long-term debt to total asset + short-term debt to total asset.

global systemically important financial institutions (G-SIFIs). Furthermore, size has been widely used to detect G-SIFIs or SIFIs for the purposes of financial regulation and supervision (IMF, BIS, and FSB, 2009; Allahrakha et al., 2015; Glasserman and Loudis, 2015). Therefore, larger FIs may have been under stricter supervision given the "too big to fail" belief prevalent since the 2008 financial crisis. This may have forced larger FIs to be more conservative in their business activities, thus reducing their potential risk spillovers. Such a mechanism surely may have occurred in China, as the government made substantial efforts to solve the problem of NPLs among the Big Four even during the late 1990s.

Fourth, we examine whether the transmission of financial shocks is affected by FI profitability as measured by net operational cash flow per share. To a certain extent, net operational cash flow per share can be considered a proxy for profitability. Better profitability will surely attract more market attention, resulting in a significant positive influence of net operational cash flow per share on total directional connectedness from others and to others, as well as net total directional connectedness during the transmission of financial shocks. For further confirmation, we also examine the influence of basic profit ratio per share, finding that it has a similar but even stronger influence pattern than net operational cash flow per share. Concerning the profit structure, we find that both financial profit and operating profit ratios negatively affect the roles of FIs in the shock transmission network. However, only the operating profit ratio is statistically significant.

Fifth, we further examine the role of FI asset tangibility. Specifically, we explore the ratios of the intangible assets to total assets and tangible assets to total assets. We document that the intangible (tangible) asset ratio negatively (positively) affects the roles of FIs in the shock transmission network (net, from, and to).

Finally, among the individually significant factors based on simple regressions, we conduct multiple regressions with Newey-West robust standard errors to select the more important factors at the 10% significance level.²¹ In the first set of multiple regressions, we include accounts receivable turnover, but not the short-term debt to total assets and liquid assets to total assets ratios due to concerns of collinearity. Although we can only obtain preliminary results from 44 observations, the findings suggest that accounts receivable turnover may not be as important as it was in the simple regression. In particular, the estimated coefficients lose their statistical significance and the adjusted R^2 turns out to be comparatively lower (Columns 1–3 of Table 8-B). Then, we conduct another set of multiple regressions while excluding accounts receivable turnover (Columns 4–6 of Table 8-B), and find that:1) The four variables that significantly affect total directional connectedness from others, to others, and the net total directional connectedness are short-term debt to total assets (negative), liquid asset to total assets (negative), size (negative), and basic profit ratio per share (positive); 2) Although each of these four factors impact directional connectedness of both from others and to others, they affect the latter ("to others") more than the former ("from others"), and thus, also affect net total directional connectedness; 3) More interestingly, compared to the macroeconomic factors presented in Table 7, firm-specific factors have much more explanatory power (adjusted R^2) with regard to the total directional connectedness to others (and thus, net total directional connectedness), while macroeconomic factors bear more influence in determining total directional connectedness from others for the transmission of financial shocks. Obviously, the above analysis is preliminary. Further research is needed to examine the issue in more depth.

7. Conclusions

This study explores the transmission of financial shocks among China's FIs using stock return data while controlling for interactions with the financial sectors of the four major global economies (i.e., the US, the UK, Germany, and Japan). Based on the newly developed network analysis, we document several novel findings of China's financial system. In particular, although banks still dominate China's financial system, nonbank FIs also bear considerable influence in the transmission network of financial shocks, thus confirming the recent growing concerns about China's shadow banking problems (e.g., Allen et al., 2012). Interestingly, the marketoriented large commercial banks played a more pronounced role than the Big Four in the financial shock transmission network during the sample period. The role that each FI plays during the transmission of financial shocks also varies over time. Furthermore, China's financial sector exerts considerable influence on financial sectors in major developed countries, especially Japan. Macroeconomic factors, especially currency-related factors, mainly determine the degree of influence from other institutions on a particular FI while firm-specific factors mainly determine the degree of a particular FI on others during the transmission of financial shocks.

The findings of this study suggest the need to reconsider the conventional approach of identifying SIFIs based on relatively lowfrequency financial data (e.g., IMF, BIS, and FSB, 2009; Allahrakha et al., 2015; Glasserman and Loudis, 2015). Such an approach could fail to capture the time-varying role that each institution may play in the transmission network of financial shocks, at least in China. The documented pattern of interconnectedness between China and the financial sectors of the other four major global economies also implies that any policy intervention in the financial sector of a major country may spill over to the financial sectors in other countries. Accordingly, some international policy coordination involving China is warranted. Finally, to achieve efficient financial regulation and supervision in China, we must be more attentive to the emerging impact of nonbank FIs (Allen et al., 2012).

Acknowledgments

The authors thank Mark Liu, an anonymous referee, session participants at the annual meetings of the 2018 China International Risk Forum and 2018 China International Conference in Finance, and seminar participants at the People's Bank of China (PBOC) for their helpful comments and suggestions. Part of the work was completed when Jian Yang was a visiting fellow and Jun Ma was the

²¹ The intangible assets to total assets ratio is excluded because of its perfect collinearity with tangible assets to total assets.

Table 8

_

Firm-specific factors.

8-A: Simple regressions

	Net			From			То		
	Estimate	Obs.	Adj-R ²	Estimate	Obs.	Adj-R ²	Estimate	Obs.	Adj-R ²
Total debt to total asset	0.054 (0.058)	576	0.003	0.014 (0.009)	576	0.012	0.068 (0.065)	576	0.005
Long-term debt to total asset	0.100* (0.054)	483	0.018	0.018* (0.009)	483	0.020	0.118* (0.061)	483	0.020
Short-term debt to total asset	-1.382*** (0.324)	483	0.061	-0.118** (0.056)	483	0.014	-1.500*** (0.377)	483	0.057
Accounts receivable turnover	-0.001*** (0.000)	144	0.150	-0.000** (0.000)	144	0.031	-0.002*** (0.000)	144	0.142
Liquidity asset to total asset	-0.739*** (0.193)	483	0.069	-0.067** (0.031)	483	0.018	-0.807*** (0.222)	483	0.065
Currency asset to total asset	0.059 (0.042)	576	0.004	0.004 (0.007)	576	-0.001	0.063 (0.047)	576	0.004
Receivable asset to total asset	0.844 (0.546)	576	0.003	0.088** (0.039)	576	0.000	0.932 (0.568)	576	0.003
Size(market value, RMB, log-)	- 0.016** (0.008)	576	0.020	-0.000 (0.001)	576	-0.001	- 0.017* (0.009)	576	0.017
Net operational cash flow per share	0.003*** (0.001)	399	0.032	0.000**** (0.000)	399	0.011	0.004*** (0.001)	399	0.030
Basic profit ratio per share	0.044*** (0.009)	576	0.036	0.007***	576	0.034	0.050*** (0.011)	576	0.039
Profit generated by financial activities to total profit	-0.003 (0.008)	576	-0.002	-0.000 (0.001)	576	-0.002	-0.003 (0.009)	576	-0.002
Operational profit to total profit	-0.250*** (0.096)	576	0.002	-0.007 (0.011)	576	-0.002	-0.257** (0.102)	576	0.002
Intangible asset to total asset	7.088**	576	0.006	0.725* (0.395)	576	0.001	7.813**	576	0.006
Tangible asset to total asset	-4.502***	576	0.019	-0.406***	576	0.005	-4.908***	576	0.018

(0.139)

(1.038)

8-B: Multiple regressions

	Dependent var	iables				
	(1)	(2)	(3)	(4)	(5)	(6)
	Net	From	То	Net	From	То
Long-term debt to total asset	-0.061	0.034	-0.027	-0.058	-0.010	-0.068
	(0.280)	(0.058)	(0.327)	(0.092)	(0.009)	(0.099)
Short-term debt to total asset				-1.079***	-0.090*	-1.169***
				(0.390)	(0.046)	(0.435)
Accounts receivable turnover	-0.001	0.000	-0.001			
	(0.001)	(0.000)	(0.001)			
Liquidity asset to total asset				-0.473***	-0.060***	-0.532***
				(0.169)	(0.018)	(0.181)
Size (market value, RMB, log-)	0.007	0.002	0.009	-0.032***	-0.003**	-0.035***
-	(0.024)	(0.004)	(0.026)	(0.009)	(0.001)	(0.010)
Net operational cash flow per share	(0.002)	(0.000)	(0.002)	(0.001)	(0.000)	(0.001)
	0.024	0.000	0.024	0.044***	0.006***	0.050***
Basic profit ratio per share	(0.016)	(0.002)	(0.016)	(0.013)	(0.002)	(0.015)
	1.316	-0.664	0.652	0.071	0.018	0.089
Operational profit to total profit	(3.504)	(0.644)	(3.960)	(0.392)	(0.043)	(0.419)
	- 4.095**	-0.323	-4.418*	-1.387	-0.007	-1.394
Tangible asset to total asset	(2.001)	(0.390)	(2.261)	(2.379)	(0.232)	(2.565)
	2.675	1.836*	4.510	2.199	1.000***	3.199
Constant	(5.109)	(0.917)	(5.783)	(2.472)	(0.242)	(2.663)
	(5.196)	(0.907)	(5.896)	(2.532)	(0.244)	(2.727)
Observations	44	44	44	306	306	306
Adj-R ²	0.128	-0.096	0.097	0.185	0.089	0.182

Notes: This table reports the results of multiple regressions with robust standard errors. The heteroscedasticity and autocorrelation consistent standard errors (HAC) are in parentheses. "*", "**", and "***" denote significance at 10%, 5%, and 1%, respectively.

(1.095)

Chief Econ /ersion of Perspectiv(upport of 53182068)	omist this I e" in v the N 1. The	at the project of the	le Res was Yand al Na ect is	earch publis long Ji tural S also su	Bure thed i a also scienc uppor	au of P n Chir partic e Four ted by	BOC, v nese as ipated idation the N	whose a PB . The t of Cl ation	e hospit 30C wc views (hina (7 al Soci:	tality orking expre- 1571 al Sci	is gree g pape ssed in 106). ¹ ence F	atly a r (No 1 this J Yu acl ?ound:	cknowl . 2017 paper a knowle ation o	ledged. -4), tit ure tho: idges fi sf Chin	. Jun l led "T se of th inanci a (118	Ma is ¿ ransm he autl al supj &ZD01	ulso a lission lission hors al port fr 6).	memb of Fir nd do 1 om th	er of th nancial not refl e Fund	le PBC Shoc ect th amen	DC mc ks anr e offic tal Re tal Re	netary ong C ial poo search	y polic lhinese sitions Fund	y com: e Finar i of PBG s for th	mittee ncial I DC. Ya he Cen	e. An e Institu ang ac itral U	earlier trions: knowl nivers	abridge A Globi edges th itties (N	o. je al
Appendix	Α																												
Table A-1 ull sample	conne	ctedn	ess ta	ble with	h US,	UK, an	d Germ	any a	s the lea	ading	market	s, 200	8-2015																
SIT	SIS	GY	S HT	Sd S.	CJE	s cs	NES	IAI	CLI	CPI	HXB	BOC	BON	CMB	IB 1	ICBC	BN F	PAB 1	MSB (CB CB	BB	B B(C PI	ISN BC	F UK	F JPF	GM	FROM	L
SIT 1C SLS 4	.8 ⁽ .3 11	4.2 5 1 6	.4 4. .6 5.	8 9 6.5	.1 5. .1 6.	4 4.5 3 5.9	9 5.3 1 6.3		5 3.7 4 3.7	3.9 3.6	3.3 2.6	2.3 2.1	3.5 3.1	2.9 2.6	3.3 2.9	2.3	3.9 3.3	3.4 2.9	2.9 2.4	2.6 2.3	2.8 2.5	3.1 2.6	2.9 2.4	2.7	0.1 0.1	0.1 0.1 0	0.3 0.3).1 89).1 89	1

~	
~	
_	
_	
u	
÷.	
-	
ω.	
Ó.	
_	
ς.	
_	
-	
Π.	
-	

GMF FROM	0.1 89	0.1 89	0.1 91	0 91	0.1 90	0.1 91	0.1 92	0.1 91	0.2 92	0.1 92	0.1 92	0.1 92	0.1 91	0.1 92	0.1 92	0.1 93	0.1 91	0 92	0 92	0.1 92	0.1 92	0.1 92	0 92	0.1 92	0.1 92	9.1 52	12.8 68	3.2 63	37.4 63	27 TC=	-36 87.3	l sectors as the	recoordinals	o, resperuvery.	
JPF	0.3	0.3	0.2	0.3	0.3	0.3	0.4	0.2	8.0.8	0.6	0.6	0.3	0.4	0.4	0.6	0.4	0.5	0.3	0.4	0.4	0.5	0.4	0.4	0.6	0.5	9.3	9.9	37.2	7.6	37	-26	nancia	todnoe	conner	
UKF	1 0.1	1 0.1	1 0.1	0	1 0.1	1 0.1	1 0.1	1 0.1	4 0.3	2 0.1	2 0.2	1 0.1	1 0.1	1 0.1	2 0.1	1 0.1	1 0.1	1 0.1	1 0.1	1 0.1	2 0.2	1 0.1	1 0.1	2 0.2	2 0.1	5 15.4	5 32.4	2 6.2	5 15.1	40	- 28	ij usm.	- unuu	counte	
USF	0.	0	0.0	0	0.0	0	0.0	0.0	0.	0	0	0	0.	0.	0.:	0.	0	0.0	0.	0.0	0.5	0	0.0		0	47.0	13.0		10.	35	- 17	ով Ger	inu cui	runun	
PDB	9 3	4 2.7	5 2.8	3.3	7 2.8	33	3.5	5 2.9	2 3.9	1 3.7	1 3.7	7 5.1	4	5.4.7	9 5.2	5.5	5.4.2	5.4.5	5.5	4.9	4.2	4.2	8 4.9	8 4.7	7.9	4 0.9	1.8	3 2.2	3 1.6	103	11	IIK a	tol div	nn m	
BC	1 2.9	5 2.	7 2.6	1 2.8	3 2.7	9 2.8	ŝ	3.2.6	8	5.4	.4	9 4.7	3.5	8.4	8.4	4.0	5.5	l 4.0	7 4.0	5	4 5.	3	9.4.6	7.7.8	6.7	5 1.	2.2	3.2.8	3 1.8	106	14	Phe IIS	not to	ווכר הח	
BB	3.	2.6	2.7	ω.	2.8	2.5	ŝ	2.8	3.6	3.6	ŝ	4	4	0.0	4.8	ß	4	5.1	4.7	4.7	4.4	4.8	7.9	4.7	4.0	0.6	1.2	1.8	1.3	101	6	ling 1	- 91111-91	nine (e	ctively
B	2.8	2.5	2.5	2.8	2.6	2.6	ŝ	2.6	3.4	3.6	3.5	4.4	4.9	4.2	4.2	4.2	4.4	4.7	4.2	4.5	4.5	8.5	4.5	4.6	3.9	0.7	1.1	1.7	1	94	7	le mor	to othe	on nate	respe
CCB	9 2.6	1 2.3	5 2.6	3 2.7	5 2.7	7 2.6	9 2.9	5 2.6	3.9	9 3.9	3.8	4.3	2 6	4.1	4.5	9 4.1	7 6.1	4	9 4.1	4.5	7 8.2	3 4.6	5 4.3	3 5.1	3 4.1	3	5 1.6	3 2.4	3 1.2	66	7	idw no	man in	· · ·	narket
MSB	2.9	2.4	2.6	2.8	2.5	2.7	2.9	2.6	3.8	3.9	3.8	5	4.7	4.4	5	4.9	4.7	4.4	4.9	8	4.7	4.8	4.6	4.8	4.8	0.8	1.6	1.8	1.3	100	8	mositic	moorto		ncial n
PAB	3.4	2.9	c	3.2	e	3.1	3.4	3.1	4.1	3.7	3.8	4.9	4.1	4.6	4.8	5.3	4.1	4.7	7.8	ß	4.3	4.6	4.7	4.5	5.2	0.7	1.2	1.9	1	103	11	decom	non long	י הוותו רו	n finai
BN	3.9	3.3	3.5	3.4	3.4	3.5	3.5	3.5	3.7	3.8	3.9	4.8	4.5	5.7	4.5	4.9	4.2	7.6	4.9	4.7	4.3	5.2	5.4	4.7	4.6	0.4	1.2	1.6	1	106	14	iance	directi	an cra	Germa
ICBC	2.3	2	2.2	2.3	2.3	2.2	2.4	2.1	3.2	3.4	3.2	3.8	5.8	3.8	4.1	3.7	9.2	3.5	3.5	4.1	5.5	4	3.7	4.6	3.6	0.8	1.3	1.9	1.2	87	4	ror vai	total	, נטנענ	and
IB	3.3	2.9	co	3.3	2.9	3.1	3.5	С	4.2	3.9	4	5.3	4.4	ß	5.3	7.5	4.5	4.9	5.6	5.3	4.5	4.8	5.3	4.8	5.7	0.8	1.6	2.1	1.5	108	15	ast er	- othor		markei
CMB	2.9	2.6	2.7	ŝ	2.8	2.7	3.2	2.7	4.3	4.1	4.1	5.3	4.7	4.6	7.7	5.2	4.9	4.4	4.9	5.2	4.8	4.7	4.9	ß	5.4	1.1	1.9	2.9	1.8	107	15	d fore	o tot n	110 I(ee	Inclal
BON	3.5	3.1	3.2	3.2	3.2	3.2	3.2	3.1	3.6	3.6	3.7	4.7	4	7.9	4.5	4.7	4.4	5.5	4.5	4.5	4.2	4.5	5.2	4.4	4.7	0.6	1.1	2	1	101	6	eralize	Cturies	unan:	se tini
BOC	2.3	2.1	2.2	2.3	2.3	2.2	2.4	2.3	3.2	3.5	3.3	3.9	9.1	3.5	3.9	3.6	5.8	3.8	3.5	4.1	5.4	4.6	3.7	4.7	3.5	0.7	1.1	1.7	0.9	86	ا ۲	ոց ներ	ן בהחוו		Japane
HXB	3.3	2.6	2.8	3.2	ĉ	ი	3.3	2.9	4	3.9	4	7.6	4.6	4.9	5.3	5.2	4.6	4.8	ß	5.3	4.7	4.9	5.1	4.9	5.3	0.5	1.3	1.7	1.2	105	13	ade-ne	-un-t-		arket, .
CPI	3.9	3.6	3.8	4	3.6	3.6	4.1	3.7	5.9	9	8.2	3.7	3.6	3.6	3.8	3.7	3.6	3.6	3.6	3.8	3.8	3.6	3.8	3.9	3.6	1	1.8	2.5	1.7	101	6	10-st	tal din	1	cial m
CLI	3.7	3.7	3.7	3.8	3.7	3.7	4.2	3.6	5.7	8.2	9	3.6	3.8	3.5	3.8	3.5	3.8	3.5	3.6	3.8	3.9	3.7	3.5	3.8	3.5	1.3	1.8	2.6	2	101	6	d from	1001 D	100° 100° 100° 100° 100° 100° 100° 100°	thnan.
PAI	3.5	3.4	3.5	3.5	3.4	3.3	3.8	3.4	8.4	5.6	5.7	3.6	3.5	3.4	3.9	3.7	3.5	3.3	3.8	3.6	3.8	3.4	3.6	3.9	3.7	2.4	2.8	3.7	2.7	102	10	len late	noctodi		iet, UK
NES	5.3	6.3	6.4	5.6	9	6.3	5.3	9.4	3.1	3.2	3.2	2.4	2.2	2.6	2.2	2.4	2.1	2.8	2.6	2.3	2.3	2.4	2.4	2.1	2.4	0.3	0.5	0.9	0.6	86	1.5	67 2240	tal con		ul mark
cs	4.9	5.9	5.8	6.6	5.9	6.2	8.3	5.9	3.9	4.1	4	3.1	2.7	3.1	ŝ	3.2	2.6	3.2	3.2	2.8	2.9	ę	2.9	2.8	3.3	0.5	0.8	1.7	0.9	66	7	actedr	intern	. יייי	nancia
CJS	5.4	6.3	6.5	6.2	6.2	8.9	5.8	6.7	3.2	3.4	3.4	2.6	2.3	2.8	2.4	2.6	2.3	ŝ	2.8	2.4	2.4	2.5	2.6	2.4	2.7	0.5	0.6	1.3	0.7	92	1	e conr	ET day		e US n
Sd	5.1	6.1	5.8	5.3	9.8	5.7	ß	5.8	2.9	3.1	3.1	2.4	2.2	2.6	2.2	2.3	2.2	2.7	2.4	2.1	2.3	2.3	2.3	2.2	2.4	0.3	0.5	1.2	0.5	83	-7	camp	n pue	ייז	ent the
HTS	4.8	5.9	5.9	8.9	5.8	6.2	6.1	5.9	3.3	3.5	3.7	2.8	2.4	2.8	2.6	2.8	2.4	2.9	2.8	2.6	2.5	2.7	2.8	2.5	2.8	0.2	0.4	1.2	0.5	91	0	he fiill		, i U	repres
GYS	2 5.4	1 6.6	5 9.2	3 5.7	4 6.2	1 6.3	4 5.2	4 6.5	5 3.2	7 3.4	5 3.4	3 2.4	7 2.2	2 2.8	3 2.3	9 2.4	7 2.2	3 2.9	1 2.6	3 2.3	7 2.3	9 2.3	9 2.4	7 2.2	2.4	4 0.5	3 0.6	9 1	3 0.7	88	- 3	norts t	Purto i	101.1 (r	GMF
SIS	3 4.2	3 11.1	5.5	4.8	7 5.4	4 5.3	3 4.4	5.4	7 2.6) 2.7	2.6	1.5	1.5	5 2.2	1.1.	3 1.5	1.7	3 2.5	5 2.1	2 1.6	1.7	3 1.5	3 1.5	1.1.5	2	4 0.≀	3 0.5	5.0 ¢	3 0.5	71	-18	hle rei	ote TC	- T - CID:	F, and
SIT	10.5	4.5	4.6	4	4.7	4.4	3.5	4.6	2.7	2.5	ŝ	2.4	2	2.5	2.1	2.5	2	2.5	2.5	2.2	2	2.5	2.5	2.1	2.2	0.4	3.0	5.0	3.0	74	- 15	This ta	a outt		KF, JP
	SIT	SLS	GYS	STH	PS	CJS	S	NES	PAI	CLJ	CPI	HXB	BOC	BON	CMB	B	ICBC	BN	PAB	MSB	CCB	CB	BB	BC	PDB	USF	UKF	JPF	GMF	TO	NET	Notes.	leading		USF, U

Ģ	Y	
	2	\$
	ç	3
	ā	5
	Ē	2
•		ť

; 2011–2015.
sectors
financial
global
major
4
and
institutions
financial
2
of 3
nnectedness
sample coi
ľIJ

Table A- Full samp	2 le connecte	sdness of 32	2 financial	institutio	ns and 4 r	najor glob	al financia	al sectors 1	2011-2015										
	SIT	SIS	GYS	HTS	PS	CJS	cs	NES	IVd	CLI	CPI	HXB	BOC	BON	CMB	IB	ICBC	BN	PAB
SIT	10	3.4	4.2	3.8	4.1	4.3	4	4.4	3	2.7	3.1	2.3	1.1	2.8	1.8	2.3	1.2	3.1	2.6
SLS	3.2	9.3	4.9	4.3	4.7	4.6	4.6	S	2.9	2.6	2.7	2	1	2.3	1.8	2.2	1.1	2.5	2.2
GYS	3.1	3.9	7.5	4.5	4.2	4.9	4.6	4.9	2.9	2.8	3.1	2.2	1	2.7	1.8	2.2	1.2	2.7	2.4
HTS	2.5	3.1	4	9.9	3.7	4.2	5.4	4	3.3	2.7	3.2	2.5	1.4	2.4	2.4	2.7	1.5	2.5	2.6
PS	3.2	3.9	4.4	4.4	7.8	4.8	4.7	4.8	5.8 0	2.5	2.7	2.2	1.1	2.4	1.9	2.2	1.2	2.7	5.3
CIS	ი ა ი	3.5 9	4.7	4.6	4.4	7.1	4.8	4.8	5.8	2.5	2.7	2.3	1.1	2.4	1.9	5.3	1.3	2.7	2.5
NEC CS	0.2	3.1	0.0 0.0	2.C	3.8 ∎	τ. Υ	0.4 4 4	4 r 0	4. c	2 u 7 c	2.5	0 C 7 C	1.4 1 -	0.7 7	2.3	1.7	c.1	0.7	1.7
DAI	7.C	6. C	4.0 2.6	4 0 0 0	c.4 4.2	9.6	0.4 0.0	6.7 9.6	6.7	C.7 5	6.4	7 5	1.1	4.7 0.0	0 T	4.5 4.6	2.2	, e	0.7 7
CLI	10	2.1	2.7	3.1	2.3	2.6	3.3	2.5	ŝ	7.4	5.5	3.2	2.5	i m	3.1	3.1	2.3	3.1	2.9
CPI	2.1	10	2.9	3.4	2.4	2.6	3.5	2.6	5.1	5.1	6.9	3.2	2.3		3.2	3.1	2.3	ę	σ
HXB	1.5	1.3	1.9	2.5	1.8	2	2.5	1.9	3.3	2.7	ŝ	6.3	ю	4	4.3	4.5	3.1	4	4.1
BOC	1	0.9	1.2	1.8	1.3	1.4	1.8	1.3	2.9	З	2.9	4.1	8.7	3.1	3.8	3.6	5.4	3.6	3.2
BON	1.8	1.7	2.4	2.5	2.1	2.3	2.6	2.2	ę	2.8	2.9	4.2	2.4	6.8	3.5	4.2	2.6	ß	3.9
CMB	1.2	1.4	1.7	2.5	1.7	1.9	2.5	1.7	3.5	2.9	3.3	4.6	3.1	3.6	6.9	4.4	3.3	3.8	3.8
IB	1.5	1.5	1.9	2.6	1.8	2.1	2.6	2	3.2	2.6	2.9	4.5	2.7	3.9	4	6.3	2.9	4.1	4.5
ICBC	1.1	1	1.4	1.9	1.4	1.6	1.9	1.4	2.9	2.7	2.9	4.1	5.3	3.3	3.9	3.8	8.5	3.4	3.3
BN	7	1.7	2.3	2.5	2.2	2.5	2.7	2.3	2.9	2.7	2.8	4.1	2.7	4.8	3.5	4.2	2.6	6.5	4
PAB	1.7	1.6	2.2	2.7	1.9	2.3	2.8	2.2	3.2	2.6	2.9	4.3	2.4	3.8	3.6	4.7	2.6	4.1	6.7
MSB	1.3	1.1	1.6	2.2	1.5	1.8	2.1	1.6	с, с с	2.9	с о о	4.8 8.0	3.1	3.7	4.5 1.5	4.9	3.2	80 · 00	4.0
CCB	1.1	1	1.6	2.2	1.5	1.7	2.1	1.7	3.1	2.8	1 ന ന	4.2	4.8 8.0	6. 0 4. 1	3.7	3.6	4.9	9.5 4.5	3.3 2
9 E	7.1	7.1	c.1	5.5	c.1	0.1 1.0	0.7 7	0.1 1	, 1 1	2 G 7 G		4 4 4	0 0 0	0.7 1	ο.υ -	4.4	0.1 0.0	4.1	<u>،</u> د
BB DG	1.0 1	1.3	9 I	2.7 2.1	1.8 1	1.2	4 C	2.1 1.2	5.2 1 C	2 C	3.1 1.0	4 r 4 c	4.7 1	4.1 7 c	4	4. ≁	6.2	4.4	4 c
DU DUR	0.1 1 4	1.1	1.0	2.6	1.0	۲.۶ ۲۰۱	2.2	۰.۱ ر	3.7	4.7 9.6	с х	4.4	4.1 2.6	0.0	0.0 6 4 3	4 4	0.0 7 C	4 °	0.0 4 7
11SF	1.1 0	0.2	0.3	0.10	50	1.7	0.4	4 0 2	4.0	0.10	0 IC	0 0 F 0	0.1	C-C	5 F O	0 ¹	0.4	0.6	4 LC
UKF	0.5	0.5	0.5	0.5	0.0	0.5	0.6	0.5		0.8	1	1.3	0.4	0.8	1.1	- 1	0.8	0.8	6.0
JPF	0.3	0.9	0.9	1	0.9	0.9	1.2	0.7	1.6	1.1	1.4	1.4	0.6	1.2	1.8	1.3	1	1.1	1.1
GMF	0.4	0.2	0.3	0.3	0.3	0.2	0.3	0.2	0.5	0.4	0.5	0.6	0.1	0.4	0.7	0.6	0.3	0.5	0.6
HuaT	2.7	3.2	4	4.6	3.8	4.2	4.6	4.1	ę	2.7	ю	2.5	1.5	2.6	2.2	2.6	1.7	2.7	2.6
GFS	2.9	3.4	4.3	4.7	3.9	4.6	4.8	4.5	3.1	2.7	°	2.3	1.2	2.4	2	2.4	1.4	2.6	2.6
CMS	2.7	3.1	3.9	4.7	3.8	4.3	4.7	4	ĉ	2.7	2.9	2.5	1.4	2.7	2.3	2.6	1.7	2.7	2.8
IS	2.8	3.6	4.2	4.6	4.5	4.7	4.6	4.5	ი	2.7	ი	2.2	1.2	2.5	1.9	2.4	1.4	2.6	2.4
ES	2.8	3.5	4.5	4.5	3.9	4.4	4.5	4.5	2.9	2.6	2.8	2.4	1.3	2.7	1.9	2.5	1.4	2.8	2.6
AB	1.2	1	1.4	1.9	1.5	1.7	2 0	1.5		2.7	8 0 7 0	4 .	5.3	3.2	80 0 10 0	3.9	5.1		0.0 0
EB C	1.3	1.2	1.0	7.7	1.7	1.8 1.8	2.3	1./	3.2	2.9	6.7	4.4	3.9	3.8	χ. α	4.2	0.5 1	4.1	3.7
10	00 L	0.5	<u></u> г,	103	ΩΩ Q	с, с	00T	76	103	41	<i>وب</i>	40T	0	101	66	108	6/	00T	102
INET	c7 –	17-	7 -	10	0	7	12	1-	10	7 -	٥	cī	c1–	ø	D	14	-12	12	ע
	MSB	CCB	CB	BB	BC	PDB	USF	UKF	JPF	GMF	H	laT C	iFS (SMS	IS	ES	AB	EB	From
SIT	1.7	1.5	1.5	2.2	2	2.1	0.1	0.2	0	.0	1	4.3	4.5	4.1	4	4.2	1.6	2	06
SLS	1.4	1.3	1.4	1.7	1.5	2.1	0.1	0.2	0.1	0		4.8	4.8	4.4	4.8	4.8	1.3	1.7	91
GYS	1.6	1.5	1.4	1.9	1.8	2.1	0.1	0.1	0.1	0	1	4.7	4.8	4.5	4.5	ß	1.4	1.8	92
HTS DC	1.9	1.9	1.8	2.1	2.1	2.5	0.1	0.1	0.1	00	1	4.8	4.7	4.7	r 4.3	4.3	1.7	2.2	93 ~~
c.IS	c.1	0'T	1.4	۲.۶ 2.1	1.7	2.2	1.0 0.1	0.1	0.1) C		4./	4.9	4.0 4.6	0 8.4	4.7 7.4	c.1	10	72 03
) S	1.8	1.8	2.7	2.1	2.1	2.5	0.1	0.1	0.1	, 0		4.7	4.7	4.6	4.2	4.3	1.7	2.2	94
NES	1.5	1.6	1.5	1.9	1.8	2.1	0.1	0.1	0.1	, O		4.8	è ю	4.5	4.7	4.9	1.5	1.9	93

J. Yang, et al.

l connected	denote <i>tota</i>	and NET of	From, TO,	mr and C.	e decompos	or variance	torecast err	lized VAR	lead genera	10-step-ah	llated from	iness calcu	connected	ull sample	sports the f	his table re	Notes: T
6	- 4	8	°	11	10	13	-12	- 46	-2	- 2	7	9	7	-11	- 4	6 	NET
102	88	101	96	104	103	107	39	11	59	46	100	66	95	81	88	89	0I
6.7	4	2.1	1.9	2.3	2.2	2.5	0	0.1	0.1	0.1	3.7	4.7	3.9	4	3.9	3.7	EB
4.6	7.7	1.9	1.7	2.1	1.9	2.2	0	0.1	0.2	0.1	3.5	4.8	3.4	3.4	5.3	3.7	AB
2.2	1.7	6.8	4.5	4.6	4.7	4.9	0.1	0.1	0.1	0.1	2.3	2.1	2.1	1.7	1.8	1.8	ES
2	1.5	4.7	7.1	4.6	4.9	4.7	0	0.1	0.2	0.1	2.2	2	2	1.6	1.6	1.7	IS
2.2	1.8	4.5	4.3	6.6	4.7	4.8	0.1	0.1	0.1	0.1	2.4	2.2	2.2	1.8	1.9	1.9	CMS
2.2	1.6	4.7	4.6	4.7	6.7	5	0.1	0.1	0.1	0.1	2.2	2	2	1.7	1.7	1.7	GFS
2.4	1.8	4.6	4.2	4.6	4.8	6.4	0	0.2	0.1	0.1	2.4	2.2	2.1	1.8	1.9	1.9	HuaT
0.2	0.3	0.3	0.2	0.4	0.3	0.2	49.2	1.8	22.3	15	0.7	0.3	0.6	0.1	0.4	0.4	GMF
1	0.8	1.1	0.9	1.1	1.1	1.2	9	42.7	9.3	6	1.2	1	1.1	0.8	0.8	0.8	JPF
0.5	0.6	0.5	0.6	0.6	0.6	0.5	17.7	2.7	39	17.7	1.2	0.7	0.9	0.5	0.8	1	UKF
0.3	0.4	0.4	0.3	0.5	0.4	0.3	13.3	1.8	20.9	49.4	0.7	0.4	0.8	0.4	0.5	0.6	USF
3.8	3.1	2.3	2.1	2.5	2.3	2.6	0.1	0.2	0.3	0.2	6.8	3.8	4	3.1	e	4	PDB
4.8	4.3	2.1	1.9	2.3	2.1	2.4	0	0.2	0.2	0.1	3.8	6.9	3.8	3.5	4.1	3.6	BC
4.2	3.2	2.2	2.1	2.4	2.2	2.3	0.1	0.1	0.2	0.2	4.2	3.9	7.1	3.4	3.2	3.7	BB
4.9	3.6	2.1	1.9	2.2	2.1	2.3	0.1	0.1	0.2	0.1	3.7	4.2	3.9	8.2	3.5	3.8	8
4.5	5.3	2.1	1.7	2.2	2	2.3	0.1	0.1	0.3	0.1	3.4	4.7	3.5	3.3	7.7	3.6	CCB
4.1	3.6	2	1.8	2.2	1.9	2.2	0.1	0.1	0.2	0.1	4.4	4	3.9	3.5	3.5	7.5	MSB
3.7	ę	2.6	2.3	2.8	2.6	2.7	0.1	0.1	0.2	0.1	4.1	3.5	3.7	3.2	2.8	3.9	PAB
4	ę	2.7	2.4	2.7	2.5	2.7	0.1	0.1	0.2	0.1	3.7	3.7	3.8	3.3	2.9	3.2	BN
4.5	5.7	1.8	1.7	2.1	1.8	2.2	0.1	0.2	0.2	0.1	3.4	4.8	3.4	3.2	5.4	3.7	ICBC
4	3.2	2.3	2.2	2.5	2.3	2.6	0.1	0.1	0.3	0.2	4.5	3.7	4	3.3	ĉ	4.1	IB
4	3.4	2	1.9	2.4	2.1	2.4	0.1	0.2	0.3	0.2	4.4	4	4	3.3	3.4	4.1	CMB
3.9	2.9	2.6	2.4	2.8	2.5	2.7	0.1	0.1	0.2	0.1	3.9	3.5	3.9	3.1	e	3.4	BON
5.1	9	1.7	1.5	1.8	1.6	2.1	0	0.1	0.1	0	3.3	5.2	3.5	4.1	5.4	3.5	BOC
4.2	3.3	2.2	2	2.4	2.2	2.5	0.1	0.2	0.3	0.2	4.2	3.9	4	3.4	3.5	4	HXB
с	2.6	2.9	ŝ	3.1	3.1	3.2	0.1	0.2	0.3	0.1	2.9	S	ŝ	2.3	2.7	2.8	CPI
3.2	2.6	2.8	2.8	ę	ę	3.2	0.1	0.2	0.2	0.1	2.9	3.1	2.9	2.5	2.7	2.9	CLI
3.2	2.6	2.8	2.8	3.1	3.2	3.2	0.1	0.2	0.3	0.2	3.1	ę	ę	2.5	2.7	2.9	IAI
	3.2 3.2 3.2 3.2 3.2 5.1 3.3 5.1 5.1 4.5 5.3 5.3 5.3 5.3 5.3 5.3 5.3 5.3 5.3 5	2.6 3.2 2.6 3.2 2.6 3.2 2.6 3.2 2.9 3.9 3.4 4 3.2 4.5 3.2 4.5 3.3 4.5 3.3 4.2 3.3 4.4 3.3 4.4 1.6 6.7 4.4 1.6 6.7 4.4 1.6 6.7 4.4 1.6 6.7 4.7 4.7 4.6 6.7 4.7 4.7 4.7 4.7 4.7 4.7 4.7 4.7 4.7 4	2.8 2.6 3.2 2.9 2.6 3.2 2.9 2.6 3.2 2.9 2.6 3.2 2.6 2.9 3.3 2.6 2.9 3.9 2.6 2.9 3.9 2.6 2.9 3.9 2.8 5.1 2.8 5.1 4.1 2.3 3.2 4.5 2.1 3.6 4.1 2.1 3.6 4.1 2.1 3.6 4.1 2.1 3.6 4.1 2.1 3.6 4.1 2.3 3.7 4.5 2.1 3.6 4.1 2.3 3.7 4.5 0.5 0.6 0.5 1.1 0.8 1 0.3 0.3 0.2 4.5 1.8 2.2 4.5 1.8 2.2 4.5 1.8 2.2 4.5 1.8 2.2 1.1 0.8 1 0.3 0.3 0.2 0.5 0.6 0.5 1.1 0.8 1 0.3 0.3 0.2 1.1 0.8 1 0.3 0.3 0.2 1.1 0.8 1 0.3 0.3 0.2 1.1 0.8 1 0.3 0.3 0.2 0.5 0.6 0.5 1.1 0.8 1 0.3 0.3 0.2 0.3 0.3 0.2 0.4 0.4 0.5 0.5 0.6 0.5 1.1 0.8 1 0.3 0.3 0.2 0.3 0.3 0.2 0.3 0.3 0.2 0.3 0.4 0.5 0.5 0.6 0.5 1.1 0.8 1 0.3 0.3 0.2 0.3 0.3 0.2 0.3 0.3 0.2 0.3 0.3 0.2 0.3 0.4 0.5 0.5 0.6 0.5	28 2.8 2.8 2.6 3.2 28 2.8 2.6 3.2 1.5 1.7 6 3.3 2.4 2.6 3.3 3.4 1.5 1.7 6 3.3 2.4 2.6 2.9 3.3 2.4 2.6 2.9 3.3 2.1 2.6 2.9 3.3 2.1 2.6 2.9 3.3 2.2 2.3 3.4 4 2.1 2.3 3.2 4.5 1.7 1.8 2.7 3.7 4.5 2.3 2.6 3 3.7 4.5 1.7 2.1 3.6 4.1 1.7 1.7 2.1 3.6 4.1 1.3 2.1 2.1 3.6 0.6 0.5 0.6 0.3 0.4 0.4 0.3 0.2 2.4 1.9 2.1 2.3 3.6 4.1 1 0.3 0.5 0.6 0.6 0.5 0.2	3.1 2.8 2.8 2.6 3.2 3.1 2.8 2.8 2.6 3.2 3.1 2.8 2.8 2.6 3.2 2.4 1.5 1.7 6 5.1 2.8 2.4 1.9 2.5 2.9 2.6 3.2 2.8 2.4 1.9 2.7 6 5.1 2.8 2.4 1.9 2.2 3.3 4.2 2.1 1.7 2.6 3.3 3.7 4.5 2.1 1.7 2.4 2.3 3.7 4.5 2.1 1.7 2.4 2.3 3.7 4.5 2.2 1.7 2.1 2.3 3.7 4.5 2.2 1.8 2.1 4.5 4.5 4.5 2.2 1.8 2.1 4.5 4.5 4.5 2.2 2.1 2.2 3.6 4.1 2.7 2.2 2.1 2.2 3.6 4.1	3.2 3.1 2.8 2.8 2.6 3.2 3.1 3.1 2.8 2.8 2.6 3.2 3.1 3.1 3.1 2.8 2.6 3.2 1.6 1.8 1.5 1.7 6 5.1 2.1 2.4 2.9 2.6 3.3 4.2 2.1 2.4 1.9 2. 3.3 4.2 2.1 2.4 1.9 2. 3.3 4.4 2.1 2.4 2.6 3.3 3.7 1.9 2.1 1.7 2.6 3.3 3.7 2.8 2.4 1.9 2.1 3.3 4.4 2.5 2.7 2.4 2.7 3.3 4.5 1.9 2.2 1.1 1.8 2.7 4.4 2.6 2.8 2.6 3.3 3.7 1.9 2.1 2.1 2.3 3.6 4.1 2.1 2.2 1.8 2.1 4.5 4.5 2.1 2.3 2.1 2.3 </td <td>32 32 31 28 2.8 2.6 32 32 31 31 2.8 2.8 2.6 32 32 31 31 3 2.8 2.6 32 21 1.6 1.8 1.5 1.7 6 5.1 21 2.5 2.8 2.4 2.6 2.9 3.3 4.2 2.7 2.5 2.8 2.4 1.9 2.6 3.3 3.7 2.6 2.3 2.5 2.1 1.7 6 5.1 4 2.7 2.5 2.8 2.4 2.7 3.3 4 4 2.7 2.5 2.1 1.9 2.1 1.8 3.7 4.5 2.7 2.6 2.8 2.6 3.3 3.7 4.5 2.7 2.6 2.3 2.1 2.3 3.6 4.1 2.7 2.6 2.3 3.6 4.1 3.8 4.5 2.3 2.1 2.3 2.1 2.3 2.6 3.7</td> <td>$\begin{array}{cccccccccccccccccccccccccccccccccccc$</td> <td>$\begin{array}{cccccccccccccccccccccccccccccccccccc$</td> <td>0.3 0.2 0.1 3.2 3.2 3.1 2.8 2.6 3.2 0.2 0.1 3.2 3.1 3.1 2.8 2.6 3.2 0.3 0.2 0.1 3.2 3.1 3.1 2.8 2.6 3.3 0.1 0.1 0.1 2.1 1.6 1.8 1.5 1.7 6 5.1 0.2 0.1 0.1 2.7 2.5 2.8 2.3 3.3 4.4 0.2 0.1 0.1 2.7 2.5 2.8 2.6 3 3.7 0.2 0.1 0.1 2.7 2.5 2.8 2.6 3 3.7 0.2 0.1 0.1 2.7 2.5 2.7 1.7 1.8 4.7 0.2 0.1 0.1 2.7 2.6 3.3 4.4 0.2 0.1 0.1 2.7 2.6 3.8 4.7 0.2 0.1</td> <td>02 03 02 01 32 32 31 28 26 32 01 02 01 32 31 31 28 26 32 01 02 01 01 1 32 31 31 28 26 33 02 01 01 01 21 1.6 1.8 1.5 1.7 6 33 02 03 02 01 01 27 25 28 24 26 33 37 44 01 02 01 01 27 25 27 24 21 21 24 49 01 02 01 01 27 25 27 24 27 37 44 01 02 01 01 23 26 3 33 45 01 02 01 27 26 23 26 36</td> <td>$\begin{array}{cccccccccccccccccccccccccccccccccccc$</td> <td>$\begin{array}{cccccccccccccccccccccccccccccccccccc$</td> <td>$\begin{array}{cccccccccccccccccccccccccccccccccccc$</td> <td>$\begin{array}{cccccccccccccccccccccccccccccccccccc$</td> <td></td> <td></td>	32 32 31 28 2.8 2.6 32 32 31 31 2.8 2.8 2.6 32 32 31 31 3 2.8 2.6 32 21 1.6 1.8 1.5 1.7 6 5.1 21 2.5 2.8 2.4 2.6 2.9 3.3 4.2 2.7 2.5 2.8 2.4 1.9 2.6 3.3 3.7 2.6 2.3 2.5 2.1 1.7 6 5.1 4 2.7 2.5 2.8 2.4 2.7 3.3 4 4 2.7 2.5 2.1 1.9 2.1 1.8 3.7 4.5 2.7 2.6 2.8 2.6 3.3 3.7 4.5 2.7 2.6 2.3 2.1 2.3 3.6 4.1 2.7 2.6 2.3 3.6 4.1 3.8 4.5 2.3 2.1 2.3 2.1 2.3 2.6 3.7	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	0.3 0.2 0.1 3.2 3.2 3.1 2.8 2.6 3.2 0.2 0.1 3.2 3.1 3.1 2.8 2.6 3.2 0.3 0.2 0.1 3.2 3.1 3.1 2.8 2.6 3.3 0.1 0.1 0.1 2.1 1.6 1.8 1.5 1.7 6 5.1 0.2 0.1 0.1 2.7 2.5 2.8 2.3 3.3 4.4 0.2 0.1 0.1 2.7 2.5 2.8 2.6 3 3.7 0.2 0.1 0.1 2.7 2.5 2.8 2.6 3 3.7 0.2 0.1 0.1 2.7 2.5 2.7 1.7 1.8 4.7 0.2 0.1 0.1 2.7 2.6 3.3 4.4 0.2 0.1 0.1 2.7 2.6 3.8 4.7 0.2 0.1	02 03 02 01 32 32 31 28 26 32 01 02 01 32 31 31 28 26 32 01 02 01 01 1 32 31 31 28 26 33 02 01 01 01 21 1.6 1.8 1.5 1.7 6 33 02 03 02 01 01 27 25 28 24 26 33 37 44 01 02 01 01 27 25 27 24 21 21 24 49 01 02 01 01 27 25 27 24 27 37 44 01 02 01 01 23 26 3 33 45 01 02 01 27 26 23 26 36	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		

Ç	5
	×
	ਰੁ
	9
	ē.
•	4

Table A-3Total directional connectedness from each sector/market (%), 2011–2015.

Trust 10	Securities	Insurance	Bank	USF	UKF	JPF			
10							GMF		
	49.3	8.8	31.8	0.1	0.2	0	0.1	68.1	0.4
3.2	61	8.2	27.5	0.1	0.2	0.1	0	72.4	0.4
S 3.1	58	8.8	29.7	0.1	0.1	0.1	0.1	69.9	0.4
S 2.5	53.8	9.2	34.2	0.1	0.1	0.1	0.1	65.5	0.4
3.2	58.1	8	29.8	0.1	0.1	0.1	0	69.3	0.3
3 3	57.6	8	31.2	0.1	0.1	0.1	0	68.6	0.3
2.6	53.2	9.4	34.4	0.1	0.1	0.1	0	65.2	0.3
S 3.2	58.5	8.2	30	0.1	0.1	0.1	0	6.69	0.3
1 2	34.1	16.1	46.7	0.2	0.3	0.2	0.1	52.2	0.8
1 2	33.4	17.9	46	0.1	0.2	0.2	0.1	53.3	0.6
I 2.1	34.7	17.1	45.4	0.1	0.3	0.2	0.1	53.9	0.7
B 1.5	25.2	6	63.8	0.2	0.3	0.2	0.1	35.7	0.8
C 1	18.4	8.8	71.6	0	0.1	0.1	0	28.2	0.2
N 1.8	28.8	8.7	60.2	0.1	0.2	0.1	0.1	39.3	0.5
IB 1.2	24.2	9.7	64.1	0.2	0.3	0.2	0.1	35.1	0.8
1.5	26.4	8.7	62.7	0.2	0.3	0.1	0.1	36.6	0.7
3C 1.1	20.2	8.5	69.7	0.1	0.2	0.2	0.1	29.8	0.6
7	29.2	8.4	60	0.1	0.2	0.1	0.1	39.6	0.5
B 1.7	28.7	8.7	60.1	0.1	0.2	0.1	0.1	39.1	0.5
B 1.3	22	9.2	60.9	0.1	0.2	0.1	0.1	32.5	0.5
B 1.1	22.1	8.9	67.3	0.1	0.3	0.1	0.1	32.1	0.6
1.2	23	8.5	66.8	0.1	0.2	0.1	0.1	32.7	0.5
1.6	24.7	9.1	64	0.2	0.2	0.1	0.1	35.4	0.6
1.3	23	6	66.1	0.1	0.2	0.2	0	33.3	0.5
B 1.4	26.4	8.6	62.6	0.2	0.3	0.2	0.1	36.4	0.8
F 0.3	4.2	1.2	8.7	49.4	20.9	1.8	13.3	5.7	85.4
F 0.5	6.3	2.8	13.3	17.7	39	2.7	17.7	9.6	77.1
ч 0.3	11.9	4.1	17	6	9.3	42.7	9	16.3	67
TF 0.4	3.2	1.4	6.8	15	22.3	1.8	49.2	5	88.3
aT 2.7	53.1	8.7	34.9	0.1	0.1	0.2	0	64.5	0.4
S 2.9	55.9	8.8	32	0.1	0.1	0.1	0.1	67.6	0.4
IS 2.7	53.4	8.6	35.1	0.1	0.1	0.1	0.1	64.7	0.4
2.8	56.7	8.7	31.2	0.1	0.2	0.1	0	68.2	0.4
2.8	55.3	8.3	33.3	0.1	0.1	0.1	0.1	66.4	0.4
1.2	20.8	8.5	68.7	0.1	0.2	0.1	0	30.5	0.4
1.3	23.5	6	66	0.1	0.1	0.1	0	33.8	0.3

	×
	8
	ž.
	ē
	5
	7
2	~

Table A-4 Full sample connected

2008-2015.
returns,
filtered
institutions'
financial
using
connectedness
ll sample (

MO																															17	ed by tional
IF FR	.1 45	.4 63	.1 70	.2 70	.2 65	.1 70	.1 69	.2 69	.4 65	.2 66	.2 65	.1 80	72	.1 75	.1 80	.1 80	73	.1 76	.1 77	.1 79	76	74	.2 78	.1 79	.1 79	.8 36	.7 51	.7 45	4 47	Ę	68	filtere
GN	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	12	21.	6	53	47	0	urns total
JPF	0.3	0	0.1	0	0	0	0	0.1	0.4	0.2	0.2	0	0	0.1	0.2	0.1	0	0	0.1	0	0.1	0	0	0.1	0.1	1	4	55.4	3.5	11	- 34	ock ret und <i>net</i> ely.
UKF	0.1	0.3	0.2	0.2	0.2	0.2	0.1	0.3	0.6	0.1	0.2	0.1	0	0	0.1	0.1	0.1	0.1	0.1	0.1	0.1	0	0	0.2	0.2	21	49.2	15.6	23.8	64	13	nns' st hers, a pectiv
USF	0.1	0	0.1	0.2	0.1	0	0	0.2	0.8	0.2	0.1	0.1	0	0	0	0	0	0.1	0	0	0.1	0	0	0.1	0	63.8	20.5	14.7	14.9	52	16	titutio to oti et, res
PDB	0.5	0.2	0.2	0.5	0.1	0.2	1.2	0.1	2.8	2.1	2	7.7	2.8	6.1	7.9	9.1	3.4	5.2	8.2	6.5	3.9	3.9	6.6	5.4	21.2	0	0.4	0.2	0.2	87	8	al ins dness marke
BC	0.4	0.4	0.2	0	0.1	0	0.1	0.2	3.4	2.9	2.9	5.5	7.8	4.8	6.1	ß	7.5	5.3	2	6.1	8	6.4	5.5	21.1	5.3	0.1	0.3	0.3	0.3	06	11	inanci nnecte ncial
BB	0.1	0.1	0.1	0.3	0	0.1	0.1	0	2.3	1.7	2.5	6.3	3.4	8.4	5.8	6.8	3.4	8	5.4	5.1	4.1	5.7	22.2	5.3	6.4	0.1	0.1	0.1	0.4	82	4	the fi <i>ral co</i> n fina
B	0.2	0.1	0	0.2	0	0	0.4	0	1.4	2.1	1.6	4.8	6.2	4.2	4	4.2	4	6.3	4.3	5.2	4.8	25.7	4.9	5.3	3.2	0.1	0.1	0.1	0	68	-6	using rection erman
CCB	0.6	0.2	0	0.1	0	0	0.2	0.1	2.7	2.8	2.2	4.2	10.6	3.2	4.5	3.4	10.7	3.1	3.4	4.8	23.9	5.2	3.7	7	3.2	0.1	0.1	0.2	0	76	0	sition <i>stal di</i> and G
MSB	0.2	0.2	0.1	0.2	0	0	0.1	0	2.4	3.3	2.8	7.1	5.1	5.1	6.9	6.7	5.1	5.1	6.7	21.3	5.4	6.2	5.3	5.9	6.3	0	0.2	0	0.1	87	8	ompo <i>vers, t</i> u urket,
PAB	0.2	0.1	0.2	0.5	0.1	0.2	0.8	0.1	3.3	2	2.1	6.2	2.7	5.1	5.7	8.2	2.9	9	22.6	6.4	3.7	4.9	5.3	4.7	7.7	0	0.1	0.2	0.1	80	ŝ	ce dec om ott- ial ma
N	0.5	0.2	0.4	0.4	0.1	0.2	0.4	0.2	1	1.3	1.5	5.3	3.6	10.1	3.9	5.6	2.6	23.6	5.7	4.6	3.2	6.8	7.5	4.6	4.7	0	0.1	0	0	75	-1	arianc tess fru inanci
CBC B	0.6	0.3	0.1	0	0.1	0.1	0.2	0.2	1.4	2.1	1.3	3.4	10	3.3	4.2	2.8	27.5	2.4	2.4	4	9.5	3.8	2.7	5.9	2.5	0	0.1	0.1	0.1	64	- 6-	error v <i>nectedn</i> anese f
B I	0.3	0.2	0.1	0.5	0.1	0.1	0.8	0	3.1	2.2	2.6	7.6	3.4	6.9	7.8	19.6	4	6.8	9.6	7.4	4.3	5.5	7.7	5.5	9.8	0	0.2	0.3	0.2	76		ecast e al con
I MB	0.5	0.2	0.1	0.2	0	0.1	0.3	0.1	4.2	3.2	3.1	7.9	4.5	5.4	20.3	7.5 1	5.8	4.7	6.4	7.4	5.4	5.1	6.3	6.5	8.2	0.1	0.3	0.7	0.3	5	5	ed for ectione marke
ON O	0.1	0.1	0.2	0.3	0.1	0.1	0.3	0	1.4	1.4	1.6	5.2	2.4	24.6	4.4	5.5	3.7	9.7	4.7	4.3	3.2	4.3	7.5	4	5.3	0	0	0.1	0.1	70	-5	talization tal dir ncial
е 0	0.7	0.2	0.1	0	0.1	0.1	0.1	0.1	1.2	5	1.2	3.1	7.9	2.1	3.2	2.4	9.8	3.1	2.2	3.9	9.1	5.7	2.7	5.8	2	0.1	0.1	0.2	0.1	-	1	ad ger tess, to IK fina
B BO	e	5	1	4			2		1	~	~	1	4	2		5	~	3	1	9	2	1		6	1		с С	1	2	9	-1	p-ahe; ectedn ket, U
ХН	3 0.	30.0	.0	~ 0.	20	2	8	6 0	6.3.	~	2	5 20.	4	1 6.	7 8	4	4	9	4	6 7.	5	2 6.	6 7	8	2	1 0	3 0.	3.0.	2	95	15	10-stej <i>l conn</i> I mar
CPI	0	0	0		0	0	1	0.0	12.	14	35	1.	1	1.	Ξ	1.	1	1	i.	1.	1.	1.	÷	1	1	0	0	0	0	53	-12	from de tota
CLI	0.2	0.6	0.8	1.5	0.6	0.6	2.5	0.5	11.6	34.4	14.3	1.5	1.7	1.1	1.9	1.3	1.7	0.9	1.4	2	2.1	1.7	1.1	2	1.3	0.1	0.2	0.2	0.3	56	-10	ulated [[] denot e US fii
IAI	0.2	0.3	0.6	1	0.3	0.2	1.5	0.4	35.1	11.2	12.6	1.6	0.9	0.9	2.3	1.7	1	0.7	2.1	1.4	1.8	1	1.3	2.1	1.7	0.1	0.4	0.6	0.4	51	- 14	ole calc nd NET sent th
NES	6.4	10	12.1	8.9	10	12.3	8.5	31.4	0.4	0.5	0.5	0	0.1	0	0.1	0	0.2	0.1	0.1	0	0	0	0	0.1	0	0.2	0.3	0.2	0.4	71	2	ess tal TO, a repre
cs	3.8	7.6	8.4	13.8	8.5	10.8	30.6	8.8	1.7	2.6	2.1	0.3	0	0.2	0.2	0.5	0	0.3	0.6	0.1	0.1	0.3	0.1	0	0.8	0	0.1	0	0	72	ŝ	ctedn ^r rom, I GMF
CJS	6.4	9.5	12.3	11.9	10.9	29.6	11.1	13.1	0.4	0.7	0.5	0	0.2	0.1	0.1	0	0.1	0.1	0.1	0	0	0	0	0	0.1	0	0.1	0	0.1	78	8	conne . TC, I F, and
PS	6.4	9.5	9.6	7.8	34.5	9.3	7.5	9.1	0.4	0.7	0.4	0	0.1	0.1	0	0	0	0.1	0	0	0	0	0	0	0.1	0	0.1	0	0.1	61	- 4	ample (Index KF, JP)
STH	4.8	8.6	10	29.9	9.2	12	14.3	9.4	1.1	1.6	2.1	0.3	0.1	0.2	0.2	0.4	0.1	0.3	0.3	0.1	0.1	0.2	0.2	0.1	0.4	0.1	0.3	0	0.1	76	9	full sé Stock SF, U
3YS I	7.5	11.2	29.8	6.6	11.1	12.1	8.5	12.6	0.7	1	1	0	0.2	0.2	0	0	0.1	0.2	0.1	0	0	0	0	0	0.1	0.1	0.2	0.2	0.3	7 7	7	s the f share shy. Ut
) SI	3.7	36.5	9.1	6.8	8.9	7.5	6.1	8.5	0.3	0.5	0.3	0	0.2	0	0	0	0.2	0	0	0	0.1	0	0	0.2	0	0	0.3	0	0.4	53	-10	treport thai A-S spective
SIT S	54.8	2.5	4	2.5	4	3.4	2.1	3.7	0	0.1	0.2	0	0.4	0.1	0.1	0	0.3	0.2	0	0	0.2	0	0	0.1	0	0.1	0.1	0.3	0	24	-21	nis table of Shang <i>lness</i> , re.
	SIT	SLS	GYS	HTS	PS	CJS	cs	NES	PAI	CLI	CPI	HXB	BOC	BON	CMB	IB	ICBC	BN	PAB	MSB	CCB	B	BB	BC	PDB	USF	UKF	JPF	GMF	TO	NET	Notes: Th returns c connected

J. Yang, et al.

References Acemoglu, D., Carvalho, V.M., Ozdaglar, A., Tahbaz-Salehi, A., 2012. The network origins of aggregate fluctuations. Econometrica 80, 1977-2016. Acemoglu, D., Ozdaglar, A., Tahbaz-Salehi, A., 2015. Systemic risk and stability in financial networks, Am, Econ, Rev. 105, 564-608. Acharya, V., Engle, R., Richardson, M., 2012. Capital shortfall: a new approach to ranking and regulating systemic risks. Am. Econ. Rev. 102, 59-64. Acharya, V.V., Pedersen, L.H., Philippon, T., Richardson, M., 2017. Measuring systemic risk. Rev. Financ. Stud. 30, 2-47. Adrian, T., Brunnermeier, M.K., 2016. CoVaR. Am. Econ. Rev. 1705-1741. Allahrakha, M., Glasserman, P., Young, H.P., 2015. Systemic importance indicators for 33 US Bank holding companies: an overview of recent data. Office of Financial Research Brief Series No. 15-01 1-7. Allen, F., Qian, J., Qian, M., 2005. Law, finance, and economic growth in China. J. Financ. Econ. 77, 57-116. Allen, F., Babus, A., Carletti, E., 2009. Financial Connections and Systemic Risk. Working Paper. University of Pennsylvania. Allen, F., Qian, J., Zhang, C., Zhao, M., 2012. China's Financial System: Opportunities and Challenges. NBER Working Paper. (No. 17828). Alpanda, S., Avsun, U., 2014. International transmission of financial shocks in an estimated DSGE model, J. Int. Money Financ, 47, 21-55. Ayyagari, M., Demirgüç-Kunt, A., Maksimovic, V., 2010. Formal versus informal finance: evidence from China. Rev. Financ. Stud. 23, 3048–3097. Bailey, W., Huang, W., Yang, Z., 2011. Bank loans with Chinese characteristics: some evidence on inside debt in a state-controlled banking system. J. Financ. Quant. Anal. 46, 1795-1830. Ballester, L., Casu, B., González-Urteaga, A., 2016. Bank fragility and contagion: evidence from the bank CDS market. J. Empir. Financ. 38 (Part A), 394-416. Benoit, S., Colletaz, G., Hurlin, C., Pérignon, C., 2013. A Theoretical and Empirical Comparison of Systemic Risk Measures. HEC Paris Research Paper No. FIN-2014-1030. Benoit, S., Colliard, J.-E., Hurlin, C., Pérignon, C., 2017. Where the risks lie: a survey on systemic risk. Review of Finance 21, 109-152. Berger, A.N., Hasan, I., Zhou, M., 2009. Bank ownership and efficiency in China: what will happen in the world's largest nation? J. Bank. Financ. 33, 113-130. Bessler, D.A., Yang, J., 2003. The structure of interdependence in international stock markets. J. Int. Money Financ. 22, 261-287. Brownlees, C., Engle, R.F., 2016. SRISK: a conditional capital shortfall measure of systemic risk. Rev. Financ. Stud. 30, 48-79. Carpenter, J.N., Whitelaw, R.F., 2017. The development of China's stock market and stakes for the global economy. Annual Review of Financial Economics 9, 233–257. Carpenter, J.N., Lu, F., Whitelaw, R.F., 2015. The Real Value of China's Stock Market. NBER Working Paper No. 20957. Cetorelli, N., Goldberg, L.S., 2012. Banking globalization and monetary transmission. J. Financ. 67, 1811-1843. Chen, X., Kim, K.A., Yao, T., Yu, T., 2010. On the predictability of Chinese stock returns. Pac. Basin Financ. J. 18, 403-425. Collin-Dufresne, P., Goldstein, R.S., Martin, J.S., 2001. The determinants of credit spread changes. J. Financ. 56, 2177-2207. Glasserman, P., Loudis, B., 2015. A Comparison of US and International Global Systemically Important Banks. Office of Financial Research Brief Series No. 15-07. Hahm, J.-H., Mishkin, F.S., Shin, H.S., Shin, K., 2012. Macroprudential Policies in Open Emerging Economies. NBER Working Paper No. 17780. Diebold, F.X., Yilmaz, K., 2014. On the network topology of variance decompositions: measuring the connectedness of financial firms. J. Econ. 182, 119-134. Hao, X., Shi, J., Yang, J., 2014. The differential impact of the bank-firm relationship on IPO underpricing: evidence from China. Pac. Basin Financ. J. 30, 207-232. Helwege, J., Zhang, G., 2015. Financial firm bankruptcy and contagion. Review of Finance 20, 1321–1362. Huang, X., Zhou, H., Zhu, H., 2009. A framework for assessing the systemic risk of major financial institutions. J. Bank. Financ. 33, 2036–2049. Huang, T.-H., Lin, C.-I., Chen, K.-C., 2017. Evaluating efficiencies of Chinese commercial banks in the context of stochastic multistage technologies. Pac. Basin Financ. J. 41, 93–110. Imai, M., Takarabe, S., 2011. Bank integration and transmission of financial shocks: evidence from Japan. Am. Econ. J. Macroecon. 3, 155-183. IMF. 2011. Global Economic Prospects and Policy Challenges. IMF, BIS, FSB, 2009. Guidance to Assess the Systematic Importance of Financial Institutions, Markets and Instruments: Initial Considerations, The Report to the G20 Finance Ministers and Governors. Jach, A., 2017. International stock market Comovement in time and scale outlined with a thick pen. J. Empir. Financ. 43, 115-129. Jarrow, R.A., Yu, F., 2001. Counterparty risk and the pricing of defaultable securities. J. Financ. 56, 1765–1799. Kamber, G., Thoenissen, C., 2013. Financial exposure and the international transmission of financial shocks. J. Money Credit Bank. 45, 127–158. Kilian, L., Lütkepohl, H., 2017. Structural Vector Autoregressive Analysis. Cambridge University Press. Lee, C.-C., Hsieh, M.-F., 2013. Beyond bank competition and profitability: can moral hazard tell us more? J. Financ. Serv. Res. 44, 87-109. Li, X., Yu, H., Fang, L., Xiong, C., October 2019. Do firm-level factors play forward-looking role for financial systemic risk: evidence from China. Pac. Basin Financ. J. 57, 101074. Mistrulli, P.E., 2011. Assessing financial contagion in the interbank market: maximum entropy versus observed interbank lending patterns. J. Bank. Financ. 35, 1114–1127. Pang, K., Siklos, P.L., 2016. Macroeconomic consequences of the real-financial nexus: imbalances and spillovers between China and the US. J. Int. Money Financ. 65, 195-212. Peek, J., Rosengren, E.S., 1997. The international transmission of financial shocks: the case of Japan. Am. Econ. Rev. 495-505. Pesaran, H.H., Shin, Y., 1998. Generalized impulse response analysis in linear multivariate models. Econ. Lett. 58, 17-29. Schnabl, P., 2012. The international transmission of bank liquidity shocks: evidence from an emerging market. J. Financ. 67, 897-932. Tobin, D., Volz, U., 2018. The Development and Transformation of the People's Republic of China's Financial System. Asian Development Bank Institute working paper No. 825. Yang, J., Zhou, Y., 2013. Credit risk spillovers among financial institutions around the global credit crisis: firm-level evidence. Manag. Sci. 59, 2343-2359. Yang, J., Hsiao, C., Li, O., Wang, Z., 2006. The emerging market crisis and stock market linkages: further evidence. J. Appl. Econ. 21, 727-744. Yang, L., van Wijnbergen, S., Qi, X., Yi, Y., 2019. Chinese shadow banking, financial regulation and effectiveness of monetary policy. Pac. Basin Financ. J. 57, 101–169.